COUNSELLORS AT LAW

David T. Doot

Direct Dial: (860) 275-0102 E-mail: dtdoot@dbh.com

January 21, 2003

VIA HAND DELIVERY

The Honorable Magalie Roman Salas Secretary Federal Energy Regulatory Commission 888 First Street Washington, D.C. 20426

New England Power Pool and ISO New England Inc. Docket No. ER02-2330-____

Report of Compliance

Dear Secretary Salas:

Re:

Pursuant to Rule 1907 of the Commission's Rules of Practice and Procedure, 18 C.F.R. § 385.1907 (2002), the New England Power Pool ("NEPOOL") Participants Committee¹ and ISO New England Inc., ("ISO-NE") hereby file an original and five (5) copies of this transmittal letter and proposed changes to NEPOOL Market Rule 1 ("Market Rule 1") to comply with the Commission's December 20, 2002 "Order On Rehearing and Accepting Compliance Filings" in the above-captioned docket. New England Power Pool, 101 FERC ¶ 61,344 (2002) (the "December 20 Order"). The Market Rule 1 changes are included in Attachment 1 of this Report. In addition to changing Market Rule 1, this Report of Compliance responds to the directives of the

¹ Capitalized terms used but not defined in this filing are intended to have the same meaning given to such terms in Section 1 of the Restated New England Power Pool Agreement, Section 1 of the Restated NEPOOL Open Access Transmission Tariff and Section 1.3 of Market Rule 1.

The Honorable Magalie Roman Salas January 21, 2003 Page 2

December 20 Order by providing further information on billing determinants proposed for the allocation of Operating Reserve Charges and by confirming the Commission's understanding regarding the allocation of Auction Revenue Rights ("ARRs") and the meaning of "Congestion Paying LSE."

As the Commission is aware, ISO-NE is currently planning to implement its Standard Market Design on March 1, 2003, with bids for the initial auction of one month of Financial Transmission Rights ("FTRs") to be submitted beginning on February 10, 2003. The ISO-NE Board intends to make a final decision on the implementation date at a February 6 meeting, following consideration of: (1) any additional guidance the Commission may provide prior to that time; (2) the results of a third market trial that was recently completed, and; (3) input received from the Participants Committee and state regulators. Once the SMD Effective Date occurs, it may be impossible to adjust settlements retroactively to effect any additional SMD compliance changes the Commission might conclude are necessary. Further, it is highly desirable for the details of the New England Load Response Program (the "Load Response Program") to be defined soon so that marketing of that Program can begin. A final Commission order on this compliance filing will provide market certainty that will enhance the success of these important market changes. Accordingly, NEPOOL and ISO-NE urge the Commission to expedite its review of this compliance filing.

I. BACKGROUND

On July 15, 2002, the NEPOOL Participants Committee and ISO-NE jointly tendered for filing Market Rule 1 and its related appendices pursuant to Section 205 of the Federal Power Act. Market Rule 1 is intended to embody a revised wholesale market design, commonly referred to in New England as the Standard Market Design ("SMD"), for the implementation of Locational Marginal Pricing ("LMP") and a multi-settlement system.

On September 20, 2002, the Commission issued its "Order Accepting In Part And Modifying In Part Standard Market Design Filing And Dismissing Compliance Filing" (the "SMD Order") in this proceeding.² In response to the SMD Order, NEPOOL and ISO-NE, as well as other parties, submitted rehearing requests and NEPOOL and ISO-NE submitted compliance filings. On December 20, 2002, the Commission issued its "Order on Rehearing and Accepting Compliance Filings." The December 20 Order required the NEPOOL Participants Committee and ISO-NE to make a number of modifications to Market Rule 1.⁴ The required changes and NEPOOL's and ISO-NE's compliance are described in Section II of this letter below, and address changes that both NEPOOL and ISO-NE believe are in full compliance with the December 20 Order. Section III of this transmittal letter provides the further information and explanations required by the

² New England Power Pool et al., 100 FERC ¶ 61,287 (2002).

³ New England Power Pool et al., 101 FERC ¶ 61,344 (2002)

⁴ <u>See Id.</u> at P 28, 45, 47, 48, 54, and 75.

The Honorable Magalie Roman Salas January 21, 2003 Page 4

December 20 Order regarding the distinctions between different methods of allocating Operating Reserve Charges, as well as a confirmation of the Commission's understandings of the ARR allocation process and the meaning of the term Congestion Paying LSEs.

At its January 10, 2003 meeting, the NEPOOL Participants Committee approved the Market Rule 1 changes reflected herein.

II. CHANGES TO MARKET RULE 1 EFFECTING COMPLIANCE

1. Level 1 Mitigation

In the December 20 Order, the Commission "reject[ed] [the] proposal for Level 1 Mitigation without prejudice to a filing that evaluates any remaining structural problems and a proposal that targets only those suppliers that obtain market power as a result of these structural problems." In compliance with the Commission's directive, Section 5.3.1 of Appendix A to Market Rule 1 has been eliminated. Conforming changes have also been made to Sections 3.1.1, 5.3.2, 5.3.3, 5.4, 5.5.2, 5.5.3, 5.7.4 and Exhibit 1 of Appendix A.

During the review of the compliance-related changes to Appendix A at the

Participants Committee meeting, concerns were expressed with the proposal to eliminate

⁵ December 20 Order at P 28.

⁶ <u>See</u> Attachment 1. Consistent with the December 20 Order, ISO-NE is currently considering various options regarding potential replacements for the "level 1 mitigation" that would target only those suppliers with market power as a result of structural problems.

The Honorable Magalie Roman Salas January 21, 2003 Page 5

Section 3.1.1 of that Appendix. That Section provides that ISO-NE will consult with Participants regarding conduct identified by ISO-NE as potentially inappropriate prior to imposing general mitigation. ISO-NE recommended that Section 3.1.1 be completely removed since it related only to the general mitigation thresholds directed to be eliminated by the December 20 Order. ISO-NE clarified and confirmed that that the removal of this Section was not intended to preclude or reduce the consultative process related to mitigation where appropriate. A motion to amend the proposed compliance changes was made to retain the language within Section 3.1.1, but modifying that Section by deleting the reference to the now inapplicable "general markets threshold" for mitigation language contained within the introductory sentence of that provision. Some objected to that proposed change as unnecessary and beyond compliance, and a change that would alter the original intent of that Section. The motion to retain a modified Section 3.1.1 failed to achieve the required support for its adoption with a 44.57% vote in favor (Attachment 4, Vote 3, tabulates this Participants Committee vote.). The original resolution to approve the compliance-related changes passed by a show of hands with no opposition and 15 abstentions.

2. Demand Response

A. Compliance Changes Unrelated to the New England Demand Response Initiative

In Paragraph 48 of the December 20 Order, in response to the Request for Clarification and/or Rehearing of the Connecticut Department of Public Utility Control,

The Honorable Magalie Roman Salas January 21, 2003 Page 6

the Commission stated that "[a]s to allocating the costs associated with the Real-Time Demand Response Program system-wide, we will allow such allocation as an initial matter, in order to encourage the development of demand response programs." As originally filed, Appendix E specified that these costs were to be allocated on a zonal basis pursuant to the applicable Load Obligation (Day-Ahead Load Obligation for the Day-Ahead Program and Real-Time Load Obligation for the Real-Time Programs).

Prior to the issuance of the December 20 Order, NEPOOL had approved different (but not un-reconcilable) changes regarding the allocation of Load Response Program costs. Those previously approved changes have been filed with the Commission pursuant to Section 205 of the Federal Power Act on December 27, 2002, in Docket No. ER03-345-000 (the "December LRP Filing"). In the December LRP Filing, NEPOOL proposed revisions to Appendix E to change the basis for allocating to Participants the costs of the Load Response Programs from Load Obligation to Network Load. The December LRP Filing is currently pending before the Commission, and NEPOOL would note that the cost allocation proposal contained in that filing is supported by the NEDRI participants.⁷

⁷ On January 15, 2003, the NEDRI participants endorsed NEPOOL's proposal to allocate demand response program costs to Network Load. The text of NEDRI's recommendation concerning cost allocation follows: "Allocate 2003 ISO RDR program costs to network load. Given the limited scale and objectives of the proposed 2003 price responsive load programs, NEDRI supports NEPOOL's proposal to allocate program costs to network load. NEDRI further supports recovery of these costs from ratepayers." The NEDRI participants approved this recommendation unanimously (PJM and the public utility commissions from New Hampshire, Massachusetts and Maine abstained).

The Honorable Magalie Roman Salas January 21, 2003 Page 7

The cost allocation pronouncements contained within paragraph 48 of the December 20 Order do not reflect the proposal contained within the December LRP Filing. Whether the Load Response Program costs are allocated locally or system-wide is a separate issue from the use of Network Load or Real-Time Load Obligation as the billing determinant. Load Response Program costs can be allocated in accordance with a Participant's share of Network Load within a particular Zone, or in accordance with a Participant's share of the entire Network Load within NEPOOL (i.e., system-wide).

The December LRP Filing explained that, due to the significant software development issues resulting from the proposed changes, ISO-NE would be unable to implement the cost allocation change to Network Load by the March 1, 2003 date currently contemplated as the SMD Effective Date. Accordingly, the December LRP Filing explained that this allocation-related change is proposed to become effective for both the Day-Ahead and Real-Time programs when the Day-Ahead Demand Response Program is implemented, as set forth in Sections 1.4 and 2.1 of Appendix E. Until that time, the costs of the Real-Time program will be allocated on the basis of Real-Time Load Obligations, as defined and quantified in Market Rule 1.

When the NEPOOL Participants Committee considered the Load Response

Program cost allocation changes directed by the December 20 Order, at least one

Participant sought changes that would mandate an allocation to Network Load as of the

SMD Effective Date (i.e., without the delay contemplated by the December LRP Filing).

The Honorable Magalie Roman Salas January 21, 2003 Page 8

ISO-NE explained in response that that the software development issues would not allow for the change requested in the near-term, and that it would be highly impractical for ISO-NE to resettle the markets after the software changes were made. In addition, some Participants viewed the "we will allow" language in Paragraph 48 as permissive rather than mandatory, ⁸ but the more widely accepted view was that the December 20 Order directed a system-wide allocation of Load Response Program costs.

⁸ In addition, during the NEPOOL discussion of the proposed compliance changes, some Participants questioned the appropriateness of allocating the costs of all of the Load Response Programs system-wide when the Commission referenced only the Real-Time Demand Response Program. When presenting these draft compliance changes to the Participants, ISO-NE explained that the logic of the Commission in revising the cost allocation for this one program should apply to the other programs as well, since they are similar in purpose and effect. For example, the Real-Time Profiled Response Program is similar to the 30-minute Real-Time Demand Response Program, but for the fact that interval meters are not required for participants in the Real-Time Profiled Response Program. Similarly, the Day-Ahead Demand Response Program is a reliability-focused program as is the Real-Time Demand Response Program. Moreover, ISO-NE expressed concern that different cost allocation methodologies for different Load Response Programs would be exceedingly burdensome on its operations. This last point is particularly noteworthy given that, consistent with the December 20 Order, ISO-NE intends to develop program revisions to allow participants in the Day-Ahead Demand Response Program whose bids are not accepted in the Day-Ahead Market to participate in the Real-Time Price Response Program. Accordingly, ISO-NE proposed that NEPOOL comply with the Commission's order by allocating all Load Response Program costs system-wide. If ISO-NE's interpretation that the Commission-directed cost allocation methodology was intended to apply to both the Real-Time and Day-Ahead components of the Load Response Program and not just the Real-Time components is incorrect, that issue can be rectified before the Day-Ahead Demand Response Program is implemented later in the year.

The Honorable Magalie Roman Salas January 21, 2003 Page 9

Following that explanation and discussion, the Participants Committee approved by a show of hands vote, with four opposing and seven Participants abstaining⁹, the changes revising Sections 1.4 and 2.3 of Appendix E.¹⁰

In addition to cost allocation issues, in Paragraph 45 of the December 20 Order, the Commission directed that the bid ceiling for the Day-Ahead Demand Response Program be raised from \$500/MWh to \$1,000/MWh. This change is included in Section 2.2 of Appendix E consistent with the December 20 Order.

⁹ The four entities in opposition are: Northeast Utilities System Companies, Central Maine Power Company, TransCanada Power Marketing Ltd., and Bangor Hydro-Electric Company. The seven entities abstaining are: LIPA, PSEG Energy Resources & Trade LLC, PG&E Energy Trading, Entergy Nuclear Generation Company, Exelon Generation Company, LLC, Virginia Electric and Power Company, and State of Maine.

¹⁰ The changes revising Sections 1.4 and 2.3 are contained on Sheet Nos. 602 and 603, which are currently pending Commission action in the December LRP Filing Docket. The compliance changes approved by NEPOOL consolidate the changes required by the December 20 Order with the changes to those sheets pending in the December LRP Filing Docket. Accordingly, Sheets 602 and 603 submitted herewith have been designated as Substitute First Revised Sheet Nos. 602 and 603, and are requested to be effective on February 23, 2003 for transactions on and after the applicable effective dates set forth in Market Rule 1 and Appendix E. Should the Commission require changes to Sheet Nos. 602 and 603 in the December LRP Filing Docket, NEPOOL will file appropriate substitute sheets to reflect the outcome of those proceedings.

The Honorable Magalie Roman Salas January 21, 2003 Page 10

B. NEDRI-related Compliance Changes¹¹

In Paragraph 47 of the December 20 Order, the Commission directed NEPOOL "to make a filing revising its demand response programs to reflect the results of the NEDRI process." The NEDRI process has, among other things, discussed and debated changes to the Load Response Programs previously filed with Commission by NEPOOL and ISO-NE as part of New England's SMD arrangements. NEPOOL understands that the process considers a broad range of proposals to enhance load response within New England, including some recommendations that are environmentally-related, retail jurisdictional or related to future considerations. ISO-NE and NEPOOL have focused on those preliminary recommendations of NEDRI contained in early draft reports as of the

http://nedri.raabassociates.org/Articles/Regional_DRChapter.1.21.doc

¹¹ The website of the New England Demand Response Initiative ("NEDRI") indicates that its function is as follows: "aimed at developing a comprehensive, coordinated set of demand response programs for the New England regional power markets. NEDRI's goal is to outline workable market rules, reliability standards, and regulatory criteria to incorporate a demand response capability into the electricity wholesale and retail markets. The Initiative will promote best practices and coordinate policy initiatives, not replace the functions that the ISO and other organizations must perform to design and implement demand-side programs."

ISO-NE believes that the NEDRI process has been useful in assembling the New England electric industry stakeholders to focus on demand response issues. ISO-NE supports *as a whole* the recommendations summarized in its Regional Demand Response chapter, which were unanimously approved by NEDRI participants attending the NEDRI plenary session on January 15, 2003 (this conflicted with a regularly scheduled NEPOOL Markets Committee meeting also held on this date). We note that some of the votes on specific issues in that chapter were dependent on adoption of the recommendations as whole. A copy the NEDRI Regional Demand Response chapter can be found at the following World Wide Web link:

The Honorable Magalie Roman Salas January 21, 2003 Page 11

date of the December 20th Order that were directed at the Load Response Programs that had been accepted by the Commission for the NEPOOL Control Area.

During the review of the preliminary NEDRI recommendations following the issuance of the December 20 Order, many Participants expressed concern and/or objection to a requirement that NEPOOL be forced to accept and implement recommendations proposed by another organization that was not party to the voluntary NEPOOL arrangements or subject to a Commission-approved decision-making process. At the same time, Participants were dedicated to complying with the Commission's directives and desired to satisfy the Commission's intentions in this regard. Accordingly, the Markets Committee reconsidered the substance of the preliminary NEDRI recommendations and again was unable to garner sufficient support for these recommendations principally because of concerns by many that the recommendations would not send the proper market signals. Nonetheless, the Participants Committee voted on January 10 with 90.56% in support (Attachment 4, Vote 2, tabulates this Participants Committee vote) to approve the changes to Appendix E that would

¹² At a high level, many Participants at the Markets Committee opposed the NEDRI recommendations out of a belief that they simply sought to increase the payments (through increases to the floor prices) made to load reductions requested pursuant to a non-market, reliability-based program otherwise inconsistent with the competitive markets designed for SMD. Concerns were also expressed that extending such a program, as recommended by NEDRI, may preclude the implementation of necessary improvements or modifications to the program. Finally, the Markets Committee did not consider NEDRI's recommended change to reduce the annual fee for non-Participants at its November 25 meeting because ISO-NE did not recommend this change as appropriate for a program being implemented at the wholesale market level.

The Honorable Magalie Roman Salas January 21, 2003 Page 12

implement the NEDRI preliminary recommendations that had been made at that time. ¹³ The approval was with the express understanding that the changes were solely to comply with the December 20 Order and without prejudice to Participants challenging the substance of those changes. Further, it was with the direction that NEPOOL also file a request for rehearing to the extent the Commission intended to require any changes beyond those approved. ¹⁴

The specific changes now before the Commission are contained in Sections 1.1, 1.3, 3.3 and 3.5 of Appendix E to Market Rule 1. In Section 1.1 of Appendix E, the annual fee assessed on non-Participants who take part in the Load Response Program has been reduced from \$5,000 per year to \$500 per year. The duration of the program has

¹³ At the Participants Committee, the debate over the appropriateness of the NEDRI-related compliance obligations culminated in a motion to amend the resolution approving such changes that would have included the NEDRI recommendations as an attachment to NEPOOL's compliance filing but not as actual changes to the Program's provisions unless and until such provisions were found to be unjust and unreasonable by the Commission. That motion to amend failed with approximately 40% of Participants voting in favor of the amendment. (Attachment 4, Vote 1, tabulates this Participants Committee vote.)

¹⁴ Subsequent to the December 20 Order, ISO-NE's development of changes to Appendix E and the Participants Committee meeting on January 10th, NEDRI did in fact include one additional recommendation concerning the Load Response Program. That recommendation concerned allowing participants in the Day-Ahead Demand Response Program to be able to also participate in the Real-time Demand Response Program (in addition to the Real-Time Price Response Program) if they qualify. This recommendation will be considered by NEPOOL and any additional changes approved as a result of that consideration will be filed with the Commission pursuant to Section 205 of the Federal Power Act.

¹⁵ See Attachment 1, Sheet Nos. 601, 602 and 604.

The Honorable Magalie Roman Salas January 21, 2003 Page 13

been increased to three years through revisions contained within Section 1.3.¹⁶ In Section 3.3, the minimum guaranteed payment to participants in the 30-minute Real-Time Demand Response Program has been changed from \$150/MWh to \$500/MWh. Similarly, the minimum guaranteed payment to participants in the 2 hour Real-Time Demand Response Program has been changed from \$100/MWh to \$350/MWh.

NEPOOL and ISO-NE request prompt Commission action on the resolution of these Load Response Program issues and would note that resolution of the current uncertainty surrounding these programs will enable program providers to fully understand and market these programs in advance of the summer peak-demand season.

3. Proxy CT Safe Harbor Bid Cap

In the December 20 Order, the Commission granted ISO-NE's request for rehearing concerning limitations on the use of the proxy CT safe harbor bid cap designed for Designated Congestion Areas ("DCAs"). These limitations on the implementation of the DCA-related Appendix A provisions were originally imposed by the Commission following its original review and acceptance of Market Rule 1. In light of the Commission's reversal on this issue, Sections 5.3.2 and 5.3.3 of Appendix A to Market

¹⁶ As noted in NEPOOL's Motion for Clarification or, in the Alternative, Rehearing, expanding the duration of the Program should not be considered a prohibition against NEPOOL and ISO-NE refining that Program based upon lessons learned.

¹⁷ See December 20th Order P 21.

The Honorable Magalie Roman Salas January 21, 2003 Page 14

Rule 1, and Section 1.2 of Exhibit 2 to that same Appendix, are being revised to reinstate the DCA provisions as originally proposed.¹⁸

4. Excepted Transactions

In Paragraph 75 of the December 20 Order, the Commission directed the deletion of new language in Section 2.1 of Appendix C that was submitted in the October 21, 2002 compliance filing. The Commission found that this language, which specified the scheduling and ARR treatment of Excepted Transactions that are also External Transactions, was an inappropriate submission in a compliance filing. In compliance with the Commission's directive, Section 2.1 of Appendix C has been revised to delete this language.¹⁹

5. Qualified Upgrade Awards

In Paragraph 54 of the December 20 Order, the Commission granted the request for clarification submitted by Duke Energy North America, LLC ("DENA") and directed ISO-NE to calculate the MW value of the increase in transfer capability created by a transmission upgrade. In addition the Commission specified that the amount of ARRs awarded should be based on this calculation. On January 14, 2003, the Markets Committee recommended that the Participants Committee approve changes to NEPOOL Manual No. 6 (Financial Transmission Rights Manual) that specifically recognize that the

¹⁸ <u>See</u> Attachment 1.

¹⁹ See Attachment 1, Sheet No. 603.

The Honorable Magalie Roman Salas January 21, 2003 Page 15

Qualified Upgrade Award calculation is in accordance with the Commission's rulings.

These changes are found in Section 8.2 of Manual 6 and can be found on ISO-NE's website. They have not yet been considered by the NEPOOL Participants Committee but are expected to be voted on at the Participants Committee February 5, 2003, meeting.

III. FURTHER INFORMATION REQUESTED BY THE COMMISSION

As noted in the introduction to the Report of Compliance, the Commission seeks further information on two issues addressed within the December 20 Order so as to confirm its understanding on ARR-related issues as well to allow it to render a decision on the proposal concerning the allocation of Operating Reserve Charges contained within Market Rule 1. Through the information and discussion contained below, NEPOOL and ISO-NE hereby respond to the Commission's directives on these issues.

1. Congestion Paying LSEs

In Paragraphs 60 and 61 of the December 20 Order the Commission states its understanding of the definition of Congestion Paying LSE and asks ISO-NE to file a statement explaining whether the Commission's understanding is correct, and if incorrect, to explain how it is incorrect.

ISO-NE states that the Commission's understanding of the term "Congestion Paying LSE" as set forth in paragraphs 60 and 61 of the December 20 Order appears to

 $^{^{20}}$ <u>See</u> http://www.iso-ne.com/committees/markets/2003_01_1415/A5_M-06_Financial% 20Transmission% 20Rights_Revison% 201.doc

The Honorable Magalie Roman Salas January 21, 2003 Page 16

be correct with the following clarification. In paragraph 60, the Commission states: "... it is our understanding that each entity serving energy to load would pay congestion costs, either by virtue of purchasing energy in ISO-NE's spot market or by paying transmission congestion charges for transmission service to move bilaterally-contracted energy to the load." Internal bilateral contracts in NEPOOL are financial in nature and do not impact the physical operation of the system. Therefore, there is no transmission service associated specifically with such transactions. An internal bilateral contract for load simply transfers the obligation to serve a load as reflected in ISO-NE's settlement system by increasing the seller's Real-Time Load Obligation and reducing the buyer's Real-Time Load Obligation at the locations specified in the internal bilateral contract submittal. As a result, the seller will be recognized in the settlement system as the entity responsible for Congestion Costs based on the Congestion Component of the locational marginal prices at the locations specified in the contract submittal to the market system.

The Congestion Paying LSE definition contains the entire set of Transmission

Customers that are eligible to receive ARRs: i.e., Transmission Customers who pay for

Regional Network Service or Long-Term Firm Point-to-Point Transmission Service

under the NEPOOL Tariff *and* who pay Congestion Costs.²¹ In addition, the definition

²¹ In the course of preparing this Compliance Filing, NEPOOL and ISO-NE have discovered that the term "Transmission Customer" has not been deleted from some places in Market Rule 1 (definition of NEMA LSE and Appendix C, Sections 1 and Section 2.2) where it should have been deleted after the change in the definition from "Congestion Paying Entity" to "Congestion Paying LSE". Only that set of Transmission Customers specified in the definition of Congestion Paying LSE is intended to receive ARRs.

The Honorable Magalie Roman Salas January 21, 2003 Page 17

includes the entities to whom those Transmission Customers transfer their obligation to supply load (in accordance with NEPOOL System Rules).

2. Real Time Load Obligation Deviation

In paragraph 92 of the December 20 Order, the Commission further directs ISO-NE and NEPOOL to file additional rationale for their decision to use Real-Time Load Obligation Deviation ("RTLOD") instead of Real Time Adjusted Load Obligation Deviation ("RTALOD") when allocating Operating Reserve charges. The Commission required ISO-NE and NEPOOL to answer three distinct questions: (1) clarify the difference between RTLOD and RTALOD; (2) distinguish between internal bilateral transactions for load and internal bilateral for energy; and (3) elaborate on the flexibility RTLOD affords generators selling to marketers.

As the Commission is aware, NEPOOL and ISO-NE have concluded that it is better to use RTLOD instead of RTALOD as part of the allocation processes of both the non-synchronized condensing Operating Reserve Charges for the Real-Time Energy Market and the Real-Time Operating Reserve Charges for Daily RMR Resources.

Section 3.2.1, as acknowledged by the Commission, defines the difference between these two billing determinants. Included as Attachment 3 to this Compliance Report is information provided by ISO-NE that responds to items 1 and 2 above. This attachment further clarifies for the Commission the difference between RTLOD and RTALOD and

NEPOOL and ISO-NE expect to make an errata filing to correct any errors found in Market Rule 1, including the correction that will remove the term "Transmission Customer(s)" from the sections specified in this footnote.

The Honorable Magalie Roman Salas January 21, 2003 Page 18

explains the distinction between internal bilateral transactions for load and internal bilateral transactions for energy.

In response to the Commission's request that NEPOOL and ISO-NE elaborate on the flexibility that RTLOD affords generators and marketers, an explanation of the source of Operating Reserve Charges is necessary to better explain why ISO-NE and NEPOOL propose to allocate the charges to RTLOD. The Operating Reserve Charges in question are uplift costs driven by the commitment of generators in the Day-Ahead Market clearing process and those committed to assure reliability pursuant to the subsequent Reserve Adequacy Analysis ("RAA") or any other subsequent unit commitment. The commitment of Resources pursuant to the RAAis typically driven by underbidding of demand (load) in the Day-Ahead Market or due to unforeseen changes in system or regional conditions. Bilateral transactions for the sale and purchase of energy in the Real-Time Energy Market (internal bilateral transactions for energy), however, are financial only and simply reflect hedges against the Real-Time Energy Market price (the Real-Time LMP, Real-Time Zonal price or Real-Time Hub price depending on the nature of Real-Time imbalance hedge sought). These financial transactions serve an important purpose in hedging supplier and load exposures to spot market prices. Accordingly, it would be both inefficient and inappropriate to allocate reliability commitment costs to these transactions. Doing so would transform the existing bilateral transactions for energy settled in the Real-Time Energy Market from an imbalance hedge to a load obligation hedge (where the buyer would be forced to buy energy imbalance plus

The Honorable Magalie Roman Salas January 21, 2003 Page 19

responsibility for synchronized reserve costs). As noted in the prior NEPOOL and ISO-NE response on this issue, internal bilateral transactions to transfer load obligations do exist. The Commission should not, however, eliminate the existing ability to enter into transactions to hedge against exposure to the Real-Time Energy Market by adopting a proposed change to the Operating Reserve Charge allocator.

Such an outcome would be inefficient for several reasons. First, it would reduce trading liquidity at a time when it is already experiencing lower activity from the current credit challenges in the industry. This additional reduction in hedging activity in the Real-Time Energy Market would occur because the scope of counterparties or their risk exposure would change. Today, generators or their long term bilateral buyers that have not sold their generation in the Day-Ahead Energy Market are able to lock in their compensation for energy imbalance in advance of Real-Time in order to reduce revenue uncertainty inherent in the Real-Time LMP price. While some of this hedging may occur in the Day Ahead Energy Market, it is quite likely that some generating capability will still face higher Real-Time dispatch levels than Resources cleared Day-Ahead. Under the Real-Time energy imbalance form of trading using the RTLOD, generators may face some risk in conducting such a sale to the extent they experience a forced outage between execution of sale and Real-Time, but they can physically hedge most other risks associated with their sale. Further, since SMD will allow such self-scheduled increases at other sites on one hour notice, even a forced outage risk can often be covered through increased generation at the Participant's other facilities. Eliminating this type of energy

The Honorable Magalie Roman Salas January 21, 2003 Page 20

imbalance transaction through the conversion to a Real-Time Adjusted Load Obligation Deviation allocator, however, would introduce greater risk and more importantly, risk that generator sellers could not physically hedge. If the RTALOD proposal were to be directed by the Commission, there would be no practical way for a supplier to avoid these charges if forced to assume them under a Real-Time energy trade. Such an outcome would require those suppliers that continued to trade energy (now energy imbalance plus Operating Reserve Charges) through a broader transaction type to reflect a premium to cover the Operating Reserve Charges and their inherent unpredictability. Today, marketers often serve as an intermediary between generator suppliers and load serving entities. This increased liquidity improves price discovery and enhances competition. NEPOOL's stakeholders fear that the allocating Operating Reserve Charges via RTALOD would both decrease supplier participation and liquidity in the Real-Time Energy Market and increase the costs to serve load through higher risk premiums required of the remaining sellers of the composite product (spot market energy plus Operating Reserve Charges). The latter problem may manifest itself through numerous layers of premiums where several marketers' trades occur between the initial purchase from a generator and ultimate sale to a load-serving entity.

Finally, as NEPOOL and ISO-NE have repeatedly stated, other transactions do exist for load serving entities who wish to hedge against the reliability-based Operating Reserve Charges. They are called internal bilateral transactions for load. There is simply no reason to require that internal bilateral transactions for energy also transfer Operating

The Honorable Magalie Roman Salas January 21, 2003 Page 21

Reserve Charge responsibility and the use of RTALOD would cause such a transfer and eliminate the primary and most important distinctions between these two types of transactions provided for under the SMD.

III. ADDITIONAL SUPPORTING INFORMATION

This filing submits those changes to NEPOOL Market Rule 1 that are required to comply with the December 20 Order. Included herewith are the following documents:

- This filing letter;
- Revised sheets of Market Rule 1 and its related Appendices (Attachment 1);
- Relevant portions of Market Rule 1 and its related Appendices marked to show the changes made since last submitted to the Commission (Attachment 2);
- Additional information provided by ISO-NE on Real Time Load Obligation Deviation and Congestion Paying LSEs (Attachment 3);
- The voting results of the Participants Committee on the changes to Market Rule 1 and its related Appendices as required by the Commission's December 20 Order (Attachment 4);
- List of NEPOOL Participants Committee members and alternates (Attachment 5);
- List of governors and utility regulatory agencies in Maine, New Hampshire, Vermont, Massachusetts, Rhode Island and Connecticut to which a copy of this filing has been sent (Attachment 6);
- A draft form of notice, suitable for publication in the <u>Federal Register</u>, and a diskette containing this form of notice, in accordance with Section 35.8 of the Commission's Regulations (Attachment 7).

Copies of this Report of Compliance and the accompanying materials are being served on all persons on the Commission's official service lists in the captioned

The Honorable Magalie Roman Salas January 21, 2003 Page 22

proceedings. Attachment 5 to this filing lists the names and addresses of each of the Participants Committee members and alternates who represent all Participants in NEPOOL. All Participants Committee members have been furnished with an electronic copy of this filing.²² A copy of this Report of Compliance has also been sent to the governors and electric utility regulatory agencies for the six New England states which comprise the NEPOOL Control Area. The names and addresses of these governors and regulatory agencies appear in Attachment 6. In accordance with the Commission's rules and practice, there is no need for the entities identified in Attachments 5 and 6 to be included on the Commission's official service lists in the captioned proceeding unless such entities already are or become intervenors in this proceeding.

Correspondence and communications regarding this filing should be addressed to the Chair of the NEPOOL Participants Committee and the undersigned as follows:

Roberto R. Denis, Chair NEPOOL Participants Committee c/o FPL Energy, Inc. 700 Universe Blvd. Juno Beach, FL 33408

Tel: (561) 691-7338 Fax: (561) 694-3650

E-mail: roberto_denis@fpl.com

David T. Doot Steven K. Steinmetz

Day, Berry & Howard LLP

CityPlace I

Hartford, CT 06103-3499

Tel: (860) 275-0266 Fax: (860) 275-0343 E-mail:dtdoot@dbh.com

sksteinmetz@dbh.com

And to the ISO as follows:

 22 Pursuant to changes to Section 21.13 (e) of the Restated NEPOOL Agreement, which was accepted by the Commission in New England Power Pool, 90 FERC ¶ 61,019 (2000), NEPOOL Participants can be served electronically rather than by hard copy.

The Honorable Magalie Roman Salas January 21, 2003 Page 23

James H. Douglass N. Beth Emery

ISO New England Inc.

Ballard Spahr Andrews & Ingersoll,
One Sullivan Road

601 13th Street, N.W. Tenth Floor

Holyoke, MA 01040-2841 Washington, DC 20005 Tel: (413) 540-4559 Tel: (202) 661-2218 Fax: (413) 535-4379 Fax: (202) 661-2299

Please acknowledge receipt of the foregoing by date-stamping and returning to the hand delivery messenger the extra copy of this filing.

Respectfully submitted, New England Power Pool Participants Committee

By: David T. Doot Its Counsel

ISO New England Inc.

By: Kathleen A. Carrigan Its General Counsel

SKS/DTD Attachments

cc: Persons designated on the Official Service List in Docket No. ER02-2330-000 Entities listed in Attachments 5 and 6

ATTACHMENT 1

Effective: January 21, 2003

APPENDIX A

MARKET MONITORING, REPORTING AND MARKET POWER MITIGATION

		Table of Contents	Page
1.	PURPOSE AND OBJECTIVES		
	1.1 1.2	ThresholdsOther Conduct	
2.	MITI	GATION PRINCIPLES	205
	2.1 2.2	Conduct Subject to Mitigation. Conditions for the Imposition of Mitigation Measures. 2.2.1 Market Impact. 2.2.2 Conduct Inconsistent with a Competitive Market.	205
	2.3	Categories of Conduct that May Warrant Mitigation	205 205 206 206
	2.4 2.5 2.6	2.3.6 Other Conduct. Other Monitoring Objectives. 2.4.1 Monitoring Targets. 2.4.2 Reports. Further Changes. Duration of Mitigation Measures.	207 207 208
3.	CON	SULTATION REQUIREMENTS	209
	3.1	In General. 3.1.1 Reserved. 3.1.2 Consideration of Information in All Cases. 3.1.3 Advance Consultation by Participant. 3.1.4 Participant Access to Its Reference Levels.	209 209 210
		J. 1.4 Faiticipalit Access to its Neterence Levels	∠ I U

Issued by: David T. Doot, Secretary

Issued on: January 21, 2003

Filed to comply with order of the Federal Energy Regulatory Commission,

Effective: January 21, 2003

		Table of Contents	Page
4.	PHY	SICAL WITHHOLDING	211
	4.1	Identification of Conduct Inconsistent with Competition	211
	4.2	Thresholds for Identifying Physical Withholding	
		4.2.1 Initial Thresholds	
		4.2.2 Adjustment to Generating Capacity	211
		4.2.3 Withholding of Transmission.	212
		4.2.4 Resources in Congestion Areas	
	4.3	Hourly Market Impact and Operating Reserve Thresholds	
	4.4	Imposition of Mitigation Measures	212
5.	ECONOMIC WITHHOLDING AND UNECONOMIC PRODUCTION		
	5.1	Purpose	213
	5.2	Applicability	
	5.3	Thresholds for Identifying Economic Withholding	213
		5.3.1 General Thresholds	
		5.3.2 Thresholds Applicable in Designated Congestion Areas	214
		5.3.3 Thresholds Applicable in Other Congestion Areas	
	5.4	Threshold for Identifying Uneconomic Production	
	5.5	Hourly Market Impact and Operating Reserve Thresholds	
		5.5.1 Initial Investigation	
		5.5.2 Reserved	
		5.5.3 Calculation of Price Impact	
	5.6	Calculation of Resource Reference Levels.	
		5.6.1 Methods for Determining Reference Levels	
	- -	5.6.2 Insufficient Data	
	5.7	Mitigation Measures5.7.1 Manual Review Prior to Mitigation	
		5.7.2 Conditions for Imposition of Mitigation Measures	
		5.7.4 Implementation	
6.	RELIABILITY MUST-RUN AGREEMENTS		
	6.1	Mitigation Agreements	220
	6.2	Cost-of-Service Agreements.	
	6.3	Filing with the Commission.	
7	МІТІ	CATION OF DEMAND RIDS	220

Issued by: David T. Doot, Secretary

Issued on: January 21, 2003

Filed to comply with order of the Federal Energy Regulatory Commission,

3. CONSULTATION REQUIREMENTS

3.1 In General. If through the application of an appropriate index or screen or other monitoring of market conditions, conduct is identified that (i) exceeds an applicable threshold, and (ii) has a material effect, as specified below, on one or more prices or Operating Reserve payments in the NEPOOL Market administered by the ISO, the ISO will take the steps set forth in this Section 3:

3.1.1 Reserved.

- **3.1.2 Consideration of Information in All Cases.** In every case, the ISO will consider all available explanations of behavior that are based on a Participant's cost of providing any market product, including
 - (a) Any relevant opportunity costs,
- (b) The need to shape bids and offers for a Limited Energy Resource to maximize the economic value from that Resource over time given the unique characteristics of the Resource, and
- (c) any special price limitations applicable to dual-fuel resources.

Issued by: David T. Doot, Secretary Issued on: January 21, 2003

Filed to comply with order of the Federal Energy Regulatory Commission,

5. ECONOMIC WITHHOLDING AND UNECONOMIC PRODUCTION

- **5.1 Purpose.** This Section defines thresholds of economic withholding in constrained areas. If conduct is detected that exceeds the thresholds specified in Sections 5.3 or 5.4 and the ISO determines that there is a market impact, as provided in Section 5.5, the conduct shall be remedied by the prospective application of a Default Offer as described in Section 5.7.
- **5.2 Applicability.** Only Resources required to offer in the Day-Ahead market will be evaluated for economic withholding in the Day-Ahead market. All Supply Offers will be evaluated in the Real-Time market.
 - 5.3 Thresholds for Identifying Economic Withholding.

5.3.1 Reserved.

Issued by: David T. Doot, Secretary Effective: January 21, 2003

Issued on: January 21, 2003

Filed to comply with order of the Federal Energy Regulatory Commission,

Effective: January 21, 2003

5.3.2 Thresholds Applicable in Designated Congestion Areas.

Pursuant to the procedures set forth in *Exhibit 2*, from time to time, the ISO may establish one or more geographic areas in which the congestion is significant and competition is limited (a "Designated Congestion Area"). Thresholds to be employed by the ISO to identify economic withholding that may warrant the mitigation of a Resource in a Designated Congestion Area are set forth in *Exhibit 2* to this *Appendix A*, and shall be determined with respect to a Reference Level determined as specified in Section 5.6, where appropriate. Resources in constrained areas that exceed the applicable thresholds and market impact tests and for which no sufficient explanation has been provided will be mitigated to the applicable Reference Level, determined in Section 5.6, unless an agreement has been negotiated under the procedures set forth in *Exhibit 2*.

- 5.3.3 Thresholds Applicable in Other Congestion Areas. For a Resource located in a constrained area that is not a Designated Congestion Area, the following thresholds shall be employed by the ISO to identify economic withholding that may warrant Mitigation Measures. Offers exceeding these thresholds and market impact thresholds and for which no sufficient explanation has been provided, shall be mitigated to the Reference Level determined as specified in Section 5.6, unless an agreement has been negotiated under the procedures set forth in *Exhibit 2*.
- (a) For Supply Offers for the Real-Time Market: for intervals in which a Resource is dispatched for the purpose of relieving a transmission constraint above the level at which it otherwise would have been dispatched ("Constrained Hours"), the ISO shall assess the market impact of any Supply Offers (Section 5.5.2(b)) that meet the following thresholds:
- (i) Energy Offer Price an increase of \$25 or 50%, whichever is lower, above the Reference Level; or; or

Issued by: David T. Doot, Secretary

Issued on: January 21, 2003

Filed to comply with order of the Federal Energy Regulatory Commission,

Effective: January 21, 2003

(ii) Start-Up or No-Load Price – an increase of 50% above the Reference Level.

- (b) For Supply Offers for the Day-Ahead Market: for all Constrained Hours (as defined above) the ISO shall assess the market impact of any Supply Offers for the Resource that meet a threshold determined in accordance with the formula specified in subsection (a).
- 5.4 Threshold for Identifying Uneconomic Production. In addition to the thresholds governing forms of economic withholding in Sections 5.3.2 and 5.3.3, the ISO will monitor for actions not consistent with competitive conduct involving uneconomic production. The following thresholds may warrant the imposition of a Mitigation Measure as provided in Section 5.7.4: (i) Energy scheduled at an LMP that is less than 20 % of the applicable Reference Level and causes or contributes to transmission congestion; or (ii) Real-Time output from a Resource that exceeds 110 % of the ISO's Dispatch Rate, and causes or contributes to transmission congestion.

5.5 Hourly Market Impact and Operating Reserve Thresholds.

- **5.5.1 Initial Investigation.** Before imposing any Mitigation Measure as permitted in Section 5.7, with regard to offers and bids identified in accordance with Sections 5.3.2 or 5.3.3, and 5.4, the ISO shall investigate the reasons for the change in accordance with the applicable provisions of Section 3. If the offers and bids in question are not explained to the satisfaction of the ISO, the ISO, in consultation with the Independent Market Advisor, will determine whether the offers and bids in question would, if not mitigated, cause a material effect on the LMP at a Node, or clearing prices in the NEPOOL Markets or Operating Reserve charges as provided in Sections 5.5.2 and 5.5.3.
- **5.5.2 Market Impact Thresholds.** Before a Mitigation Measure is imposed on offers exceeding the conduct thresholds, the ISO will determine whether there is an impact as follows:
 - (a) Reserved.

Issued by: David T. Doot, Secretary

Issued on: January 21, 2003

Filed to comply with order of the Federal Energy Regulatory Commission,

(b) For offers exceeding thresholds in Section 5.3.2, a material effect is one in excess of the threshold specified in *Exhibit 2*, Section 2.4, or Operating Reserve payment thresholds as specified in *Exhibit 1*, Section 2. For offers exceeding thresholds in Section 5.3.3, a material effect is one in excess of the conduct threshold specified in Section 5.3.3 above or Operating Reserve payment thresholds as specified in *Exhibit 1*, Section 2.

5.5.3 Calculation of Price Impact.

- (a) When it has the capability to do so, the Market Monitoring Unit, in consultation with the Independent Market Advisor, shall determine the effect on prices or Operating Reserve payments of questioned conduct through the use of sensitivity analyses performed using the ISO's unit dispatch system software ("UDS"), unit commitment software ("UCS"), scheduling, pricing and dispatch software ("SPD") and such other computer modeling or analytic methods as the ISO, in consultation with the Independent Market Advisor, shall deem appropriate.
- (b) When a determination in accordance with paragraph (a) above is not practicable, including, but not limited to when market operations are being performed in the back-up control center during an Emergency, the ISO, in consultation with the Independent Market Advisor, shall determine the effect on prices or Operating Reserve payments of questioned conduct using the best available data and such models and methods as they shall deem appropriate. The price impact analysis will be performed to allow *ex ante* mitigation in the Day-Ahead Energy Market. *Ex ante* mitigation in the Real-Time Energy Market will be performed as soon as practicable.
- (c) The Market Monitoring Unit may, in consultation with the Independent Market Advisor, set thresholds below which it need not apply the UDS, UCS, SPD, and other systems if it is reasonable to conclude that the market impact thresholds are not likely to be violated.
- (d) In constrained areas, if appropriate models are not available as the result of limitations in hardware, software, or other technical difficulties, the ISO will manually evaluate the impact to determine if it is at least as large as the threshold value. If that is not practicable, then either of the following will be deemed to be a violation of the market impact screen for a constrained area Resource exceeding a conduct threshold specified in Section 5.3: (i) the scheduling of such

Issued by: David T. Doot, Secretary Effective: January 21, 2003

Issued on: January 21, 2003

Filed to comply with order of the Federal Energy Regulatory Commission,

Effective: January 21, 2003

- (b) The market impact thresholds described in Section 5.5 are exceeded.
- **5.7.3 Level of Default Offers.** A substitute mitigated offer (a "Default Offer") shall be designed to cause a Participant to offer as if it faced workable competition during a period when (i) the Participant does not face workable competition, and (ii) has responded to such condition by engaging in the physical or economic withholding. In designing and implementing Default Offers, the ISO shall seek to avoid causing a Resource to offer below its marginal cost.

5.7.4 Implementation.

- (a) The Default Offer may establish a mitigated value for one or more components of the offer for a given Resource equal to a Reference Level for that component of the Resource's offer determined as specified in Section 5.6.1.
- (b) A Resource subject to a Default Offer shall be paid the LMP or other market clearing price applicable to the output from the Resource. Accordingly, a Default Offer shall not limit the price that a Resource may receive or pay unless the Default Offer determines the LMP or other market clearing price applicable to that Resource.
- (c) Mitigation Measures shall not be applied if the price effects of the measures would cause the average day-ahead energy price in the mitigated locations or zones to rise over the entire day.
- (d) When Mitigation Measures are applied to Offer Data, mitigation shall be imposed from the first hour in which the impact test is met to the last hour in which the impact test is met, or for the duration of the mitigated Resource's minimum run time, whichever is longer.
- (e) The posting of the Day-Ahead schedule, rebidding period and reliability commitment run may be delayed if necessary for the completion of mitigation procedures.
- (f) Mitigation that does not affect the LMP or a clearing price in another ISO market may be applied in the settlement process.

Issued by: David T. Doot, Secretary

Issued on: January 21, 2003

Effective: January 21, 2003

EXHIBIT 1

MARKET IMPACT AND OPERATING RESERVE THRESHOLDS

1. RESERVED

2. OPERATING RESERVE THRESHOLD

An increase of more than 100 % in Operating Reserve Credits due to the Participant facing mitigation in a dispatch day, provided that the increase also exceeds \$10/MWh, compared to the Operating Reserve Credits calculated using Reference Levels as determined in Section 4.5 and the physical bid characteristics for the Resource. This calculation is as follows:

$$OR_e = StartupPrice + \sum_t [NoLoadPrice_t + (SE_t \times EBB_t) - (SE_t \times LMP_t)]$$

Where:

OR_e = Operating Reserve Energy Market Component

StartupPrice = Bid Startup Price (or Reference Level)

NoLoad Price = Bid No-Load Price (or Reference Level)

SE = Supplied Energy (or Reference Economic Minimum)

EBB = Energy Bid Block Prices (or Reference Levels)

LMP = Locational Marginal Price

Issued by: David T. Doot, Secretary

Issued on: January 21, 2003

Filed to comply with order of the Federal Energy Regulatory Commission,

 Operating Hour of the Resource associated with one continuous start-up/dispatch period when Energy was Supplied (or as determined by Reference Characteristics)

The ISO, in consultation with the Independent Market Advisor, shall determine the effect of questioned conduct on Operating Reserve Charges using the best available data and such models and methods, as it deems appropriate.

t

Notwithstanding the foregoing, the ISO may determine additional RMR Congestion Areas at any time on an emergency basis or as the result of new or changed circumstances; with such prior notice and opportunity for comment as is practicable under the circumstances.

Mitigation of Resources Within Designated Congestion Areas. So 1.2 long as Supply Offers for Resources within the Designated Congestion Area are below the applicable Designated Congestion Area Threshold, no other congestion area Mitigation Measures will apply. Offers exceeding the applicable Designated Congestion Area Threshold that fail the market impact threshold test specified in Section 2.4 of this Exhibit 2 will be mitigated to the applicable Reference Level as provided in Appendix A, Section 5.6. Notwithstanding the foregoing, Resources with Supply Offers and Reference Prices exceeding the Designated Congestion Area Threshold will only be evaluated for mitigation if they exceed the market impact threshold test specified in Section 2.4 of this Exhibit 2.

2. DESIGNATED CONGESTION AREA THRESHOLD DETERMINATION

Threshold Calculation. Annually, or more frequently as the ISO deems necessary, the ISO will calculate a threshold for the upcoming 12 calendar months (the "Designated Congestion Area Threshold"). Supply Offers below the threshold will not be mitigated unless they violate the thresholds applicable in all areas, e.g., Section 5.3.1 of Appendix A. The Designated Congestion Area Threshold will be calculated as follows:

Designated Congestion Incremental Proxy Net Annual Fixed Cost CT Operating Cost + Annual Constrained Hours Area Threshold

Where:

Incremental Proxy = Calculated as specified in 5.6.1(B)(iii) of Appendix A CT Operating Cost

for a combustion turbine with a 10,500 heat rate, using a 3 month average of the 12 monthly forward gas prices plus transportation charges

Effective: January 21, 2003

Issued by: David T. Doot, Secretary

Issued on: January 21, 2003

Filed to comply with order of the Federal Energy Regulatory Commission,

First Stage ARR Allocation

AUCTION REVENUE RIGHTS AND QUALIFIED UPGRADE AWARDS

1. INTRODUCTION

Auction Revenue Rights ("ARRs") are rights to receive FTR Auction Revenues from the sale of FTRs other than FTRs sold by FTR Holders. Qualified Upgrade Awards are rights to receive FTR Auction Revenues as provided in Section 1. ARRs shall be determined and allocated to Congestion Paying LSEs, Transmission Customers and NEMA LSEs (including any of the foregoing that are parties to Excepted Transactions that are included in the list of transactions in Attachments G and G-2 of the NEPOOL Tariff), using a four-stage process as described below (the "ARR Allocation").

2. FIRST STAGE ARR ALLOCATION

2.1 **Excepted Transactions.** In the first stage of each ARR Allocation, each entity serving load to which energy is delivered or making an External Transaction sale pursuant to an Excepted Transaction included in the list of transactions in Attachments G and G-2 of the NEPOOL Tariff, and which is the party responsible for paying Congestion Cost associated with energy purchased under the Excepted Transaction shall have the option to be allocated ARRs from the generator to the location of the load or External Node. Alternatively, each seller delivering energy pursuant to an Excepted Transaction to an entity serving load or making an External Transaction sale and which seller is the party responsible for paying Congestion Cost associated with energy purchased under the Excepted Transaction shall have the option to be allocated ARRs from the generation source to the location of the load or External Node. If the party responsible for paying the Congestion Cost associated with energy purchased under the Excepted Transaction does not elect to be allocated ARRs under Section 2.1, then the ARRs associated with the destination Node(s) of the load served by such Excepted Transaction shall be allocated pursuant to Section 2.2.

2.1.1 Requesting Allocation of First Stage ARRs for Excepted Transactions. In order to be eligible to receive ARRs in association with an Excepted Transaction, each entity to which energy is delivered pursuant to an Excepted Transaction or which delivers energy pursuant to an Excepted Transaction must request that it be allocated ARRs pursuant to this Section 2.1 and in accordance with the NEPOOL Manuals and ISO Administrative Procedures prior to the first stage of the ARR Allocation for an FTR Auction.

Issued by: David T. Doot, Secretary Effective: January 21, 2003

Issued on: January 21, 2003

Filed to comply with order of the Federal Energy Regulatory Commission,

Effective: January 21, 2003

Introduction

APPENDIX E

LOAD RESPONSE PROGRAM

1. INTRODUCTION

- 1.1 Goal. The purpose of the Load Response Program ("LRP") is to facilitate load response during periods of peak electricity demand by providing appropriate incentives. Load Response Program incentives are available to any Participant or Non-Participant which, consistent with the requirements set forth herein, enrolls itself and/or one or more retail customers ("Demand Resources") to provide a reduction in their electricity consumption in the NEPOOL Control Area during peak demand periods. Non-Participants that wish to participate in the Load Response Program and have satisfied the applicable financial assurance criteria will be charged an annual service fee of \$500. The service fee will be applied to ISO expenses and may be superceded by a future provision in the ISO Tariff.
- **1.2 Eligibility.** The overall Load Response Program comprises the following individual components:
 - Day-Ahead Demand Response Program
 - Real-Time 30 Minute Demand Response Program
 - Real-Time 2 Hour Demand Response Program
 - Real-Time Price Response Program
 - Real-Time Profiled Response Program

These programs are further defined in this Appendix E and the NEPOOL Manuals.

Demand Resources are only eligible to participate in one program at a time, except that a Demand Resource participating in the Day-Ahead Demand Response Program whose offer is not accepted in the Day-Ahead Energy Market, may participate in the Real-Time Price Response Program. Generating Resources that are already qualified as generating assets are not eligible to participate in the Load Response Program.

Any Demand Resource that is eligible to qualify as an ICAP Resource must be available, at a minimum, for the summer period (May 1 through September 30).

Issued by: David T. Doot, Secretary

Issued on: January 21, 2003

Effective: February 25, 2003

Introduction

- **1.3 Effectiveness.** The Load Response Program will be effective from the SMD Effective Date through February 28, 2006; except as provided in Section 2.1.
- **Allocation of Costs.** The costs of Real-Time Load Response Programs will be allocated to the applicable Real-Time Load Obligation on a system wide basis (commencing on the SMD Effective Date), except for the costs associated with the Internet-based Communication System (IBCS) as provided in Section 7. Until the Day-Ahead program is implemented, Real-Time program costs will be allocated to Real-Time Load Obligation on a system wide basis (calculated by summing the Real-Time Load Obligation in each Load Zone). Commencing on the date that the Day-Ahead program is implemented, the allocation of the Load Response Program costs (except for IBCS costs provided in Section 7) will change from Load Obligation to Network Load on a system wide basis. To the extent that a program participant's bid in the Day-Ahead Demand Response Program clears (is accepted), any charges or credits associated with such deviations will be allocated to the program participant. The balancing credit or charge will be allocated to Network Load on a system wide basis. As stated in Section 2.1, the effective date for the Day-Ahead Demand Response Program will be the date specified by the ISO and posted on its website. Such date will be at least two weeks after the ISO has given the Commission written notice and has posted on its website that the NEPOOL System Rules and computer programs necessary to implement the Day-Ahead Demand Response Program are fully in place and functional.
- **1.5 General Requirements.** A Demand Resource cannot span multiple Load Zones. A Price Response customer cannot span multiple Load Zones. All programs, except the Profiled Response Program, require interval metering. With the exception of the Profiled Response Program and "Super" Low-Tech option of the Real-Time Price Response Program, meters are read at least daily and some will require an Internet-based Communication System.

Issued by: David T. Doot, Secretary

Issued on: January 21, 2003

Filed to comply with order of the Federal Energy Regulatory Commission,

Docket No. ER02-2330-001 et al., issued December 20, 2002, 101 FERC ¶ 61,344 (2002).

2. DAY-AHEAD DEMAND RESPONSE PROGRAM

Demand Resources that require more than 2 hours advance notice in order to curtail consumption may participate in the Day-Ahead Demand Response Program. Demand Resources with response times less than 2 hours may also participate in the Day-Ahead Demand Response Program. The Day-Ahead Demand Response Program is not intended to pay for load reductions that would have been scheduled in any event, such as facility shut-downs.

- **2.1 Effective Date.** The Day-Ahead Demand Response Program will be effective as soon as practicable, but no earlier than the SMD Effective Date. The effective date for this program will be the date specified by the ISO and posted on its website. Such date will be at least two weeks after the ISO has given the Commission written notice that the NEPOOL System Rules and computer programs necessary to implement the Day-Ahead Demand Response Program are fully in place and functional and the ISO will post the date on its website at the time that the ISO makes such notice to the Commission.
- **2.2 Offer Parameters.** A program participant may submit Supply Offers in the Day-Ahead Energy Market on behalf of a Demand Resource in increments of 1 MW or more. Resources may be aggregated to reach the 1 MW minimum. The minimum Supply Offer shall be \$50/MWh and the maximum shall be \$1,000/MWh. Demand Resources that participate in the Day-Ahead Demand Response Program are eligible to qualify as an ICAP Resource subject to the performance criteria identified within the NEPOOL Manuals.
- **2.3 Payment.** Demand Resource Supply Offers that clear the Day-Ahead Energy Market will be paid the applicable Day-Ahead Zonal Price. To the extent a program participant's bid in Day-Ahead Demand Response Program clears (is accepted), and the participant's Real-Time load response deviates from its nominated response, any charges or credits associated with such deviations will be allocated to the program participant. The balancing credit or charge will be allocated to Network Load on a system wide basis. Data for calculating actual performance, including the base line and actual reductions shall be provided on a daily basis with other meter reading data.

Issued by: David T. Doot, Secretary Effective: February 25, 2003

Issued on: January 21, 2003

3. REAL-TIME DEMAND RESPONSE PROGRAMS

- **3.1 General Terms.** The Load Response Program includes two components that provide payments for Demand Resources that are willing and capable of responding to Real-Time ISO instructions to interrupt load within a specific time period. The minimum, aggregated size to participate in these programs is 100 kW.
- **3.1.1 Technical Requirements.** Both the 30-Minute Demand Response and the 2-hour Demand Response Programs require the use of an Internet-based Communication System.
- **3.1.2 Program Activation.** The ISO may issue interruption instructions to Demand Resources on a zonal or system wide (implemented in blocks) basis. The ISO may issue interruption instructions in blocks and not just by zone, to allow for a controlled implementation. A block is a system wide slice of the Demand Resources (approximately 200 MW per block).
- **3.2 30 Minute Demand Response Program.** The 30 Minute Demand Response Program requires a Demand Resource to respond within 30 minutes of the ISO's instructions to interrupt.
- **3.3** Payment **30** Minute Demand Response Program. Program participants receive the higher of the applicable Real-Time Zonal Price for interrupted consumption (measured against the base line) or a guaranteed minimum payment of \$500/MWh for a minimum of 2 hours. Demand Resources that participate in the 30 Minute Demand Response Program are eligible to qualify as an ICAP Resource, subject to the performance criteria identified in the NEPOOL Manuals.
- **3.4 2 Hour Demand Response Program.** The 2 Hour Demand Response Program requires a Demand Resource to respond within 2 hours of the ISO's instructions to interrupt.
- **3.5** Payment 2 Hour Demand Response Program. Program participants will receive the higher of the applicable Real-Time Zonal Price for interrupted consumption (measured against the base line) or a guaranteed minimum payment of \$350/MWh for a minimum of 2 hours. Demand Resources that participate in the 2 Hour Demand Response Program are eligible to qualify as an ICAP Resource, subject to the performance criteria identified in the NEPOOL Manuals.

Issued by: David T. Doot, Secretary

Issued on: January 21, 2003

Filed to comply with order of the Federal Energy Regulatory Commission,

Docket No. ER02-2330-001 et al., issued December 20, 2002, 101 FERC ¶ 61,344 (2002).

ATTACHMENT 2

Table of Contents

APPENDIX A

MARKET MONITORING, REPORTING AND MARKET POWER MITIGATION

		Table of Contents	Page
1.	PUR	POSE AND OBJECTIVES	204
	1.1 1.2	Thresholds. Other Conduct	
2.	MITI	GATION PRINCIPLES	205
	2.1 2.2	Conduct Subject to Mitigation. Conditions for the Imposition of Mitigation Measures. 2.2.1 Market Impact.	205 205
	2.3	 2.2.2 Conduct Inconsistent with a Competitive Market. Categories of Conduct that May Warrant Mitigation. 2.3.1 In General. 2.3.2 Conduct Not Deemed Anti-Competitive. 2.3.3 Physical Withholding by a Participant. 2.3.4 Other Withholding of Supply Offers. 2.3.5 Anti-Competitive Increment Offers and Decrement Bids. 	205 205 206 206
	2.4 2.5 2.6	2.3.6 Other Conduct	207 207 208 208
3.	CON	SULTATION REQUIREMENTS	209
	3.1	In General	209 209 210

Issued by: David T. Doot, Secretary Effective: September 15, 2002 January 21, 2003

Issued on: July 15, 2002 January 21, 2003

Filed to comply with order of the Federal Energy Regulatory Commission,

Docket No. ER02-2330-001 et al., issued December 20, 2002, 101 FERC ¶ 61,344 (2002).

		Table of Contents	Page
4.	PHY	SICAL WITHHOLDING	211
	4.1	Identification of Conduct Inconsistent with Competition	211
	4.2	Thresholds for Identifying Physical Withholding	211
		4.2.1 Initial Thresholds	
		4.2.2 Adjustment to Generating Capacity	
		4.2.3 Withholding of Transmission.	
		4.2.4 Resources in Congestion Areas	
	4.3	Hourly Market Impact and Operating Reserve Thresholds	
	4.4	Imposition of Mitigation Measures	212
5.	ECO	NOMIC WITHHOLDING AND UNECONOMIC PRODUCTION	213
	5.1	Purpose	213
	5.2	Applicability	
5.	5.3	Thresholds for Identifying Economic Withholding	
		5.3.1 General Thresholds	213
		5.3.2 Additional Thresholds Applicable in Designated Congestion Areas. 214	
		5.3.3 Additional Thresholds Applicable in Other Congestion Areas.	214
	5.4	Threshold for Identifying Uneconomic Production	
	5.4 5.5	Hourly Market Impact and Operating Reserve Thresholds	
		5.5.1 Initial Investigation	
		5.5.2 Market Impact Thresholds Reserved	
	5.0	5.5.3 Calculation of Price Impact	
	5.6	Calculation of Resource Reference Levels.	
		5.6.1 Methods for Determining Reference Levels	
	5.7	Mitigation Measures	
	5.7	5.7.1 Manual Review Prior to Mitigation.	
		5.7.2 Conditions for Imposition of Mitigation Measures	
		5.7.3 Level of Default Offers	
		5.7.4 Implementation	
6.	RELI	IABILITY MUST-RUN AGREEMENTS	220
	6.1	Mitigation Agreements.	220
	6.2	Cost-of-Service Agreements	220
	6.3	Filing with the Commission.	220
7.	MITI	GATION OF DEMAND BIDS	220

Issued by: David T. Doot, Secretary
Issued on: July 15, 2002 January 21, 2003 Effective: September 15, 2002 January 21, 2003

3. CONSULTATION REQUIREMENTS

- **3.1 In General.** If through the application of an appropriate index or screen or other monitoring of market conditions, conduct is identified that (i) exceeds an applicable threshold, and (ii) has a material effect, as specified below, on one or more prices or Operating Reserve payments in the NEPOOL Market administered by the ISO, the ISO will take the steps set forth in this Section 3:
- **3.1.1** Notice and Opportunity to RespondReserved. Before imposing mitigation for violation of general market thresholds (excluding thresholds regarding congestion mitigation),
- (a) the ISO will, whenever practicable, contact the Participant engaging in the identified conduct to request an explanation of the conduct;
- (b) If the explanation, if available, considered together with other information available to the ISO, indicates to the satisfaction of the ISO that the questioned conduct is consistent with competitive behavior, no further action will be taken; and
- (c) The ISO will consider any information a Participant submits, but is not required to delay mitigation while waiting for information.
- **3.1.2 Consideration of Information in All Cases.** In every case, the ISO will consider all available explanations of behavior that are based on a Participant's cost of providing any market product, including
 - (a) Any relevant opportunity costs,
- (b) The need to shape bids and offers for a Limited Energy Resource to maximize the economic value from that Resource over time given the unique characteristics of the Resource, and
- (c) any special price limitations applicable to dual-fuel resources.

Issued by: David T. Doot, Secretary Effective: September 15, 2002 January 21, 2003

Issued on: July 15, 2002 January 21, 2003

5. ECONOMIC WITHHOLDING AND UNECONOMIC PRODUCTION

- **5.1 Purpose.** This Section defines thresholds of economic withholding in the unconstrained market and in constrained areas. If conduct is detected that exceeds the thresholds specified in Sections 5.3 or 5.4 and the ISO determines that there is a market impact, as provided in Section 5.5, the conduct shall be remedied by the prospective application of a Default Offer as described in Section 5.7.
- **5.2 Applicability.** Only Resources required to offer in the Day-Ahead market will be evaluated for economic withholding in the Day-Ahead market. All Supply Offers will be evaluated in the Real-Time market.

5.3 Thresholds for Identifying Economic Withholding.

- **5.3.1 General Thresholds**Reserved. The ISO shall investigate the reasons for and market impact of any offers that exceed the following thresholds. Offers exceeding these thresholds and market impact thresholds and for which no sufficient explanation has been provided, shall be mitigated to the Default Offer as determined in Section 5.7.
- (a) Energy Offer Price. A 300 % increase or an increase of \$100 per MWh above the Reference Level, whichever is lower, but excluding offers under \$25;
- (b) Startup and No-load Offer Price. A 200 % increase above the Reference Level.
- (c) Regulation Offers. A 300 % increase or an increase of \$25 per MW of Regulation above the Reference Level, whichever is lower, but excluding bids under \$5.
- (d) Time Based Offer Parameters. An increase greater than 2 hours in elements of a generation Resource's Offer Data that are expressed in time (e.g., minimum run time, minimum down time, cold start time, hot start time) or greater than six hours for any combination of such time-based Offer Data compared to the unit's Reference Levels.

Issued by: David T. Doot, Secretary Effective: September 15, 2002 January 21, 2003

Issued on: July 15, 2002 January 21, 2003

5.3.2 (e) Offer Parameters Expressed Other than in Time or Dollars. A 100 % increase for Offer Data that are minimum values, or a 50 % decrease for Offer Data that are maximum values (including, but not limited to, ramp rates and maximum starts per day). Additional Thresholds Applicable in Designated Congestion Areas. Pursuant to the procedures set forth in Exhibit 2, from time to time, the ISO may establish one or more geographic areas in which the congestion is significant and competition is limited (a "Designated Congestion Area"). Thresholds in addition to those set forth in Section 5.3.1 to be employed by the ISO to identify economic withholding that may warrant the mitigation of a Resource in a Designated Congestion Area are set forth in *Exhibit 2* to this *Appendix A*, and shall be determined with respect to a Reference Level determined as specified in Section 5.6, where appropriate. Resources in constrained areas that exceed the applicable thresholds and market impact tests and for which no sufficient explanation has been provided will be mitigated to the applicable Reference Level, determined in Section 5.6, unless an agreement has been negotiated under the procedures set forth in *Exhibit 2*. The additional thresholds established in Exhibit 2 for Designated Congestion Areas shall be applied only when transmission constraints and demand conditions in the Designated Congestion Area require the dispatch of all capacity of all available resources within such area.

5.3.3 Additional Thresholds Applicable in Other Congestion Areas. In addition to the thresholds set forth in Section 5.3.1, for For a Resource located in a constrained area that is not a Designated Congestion Area, the following thresholds shall be employed by the ISO to identify economic withholding that may warrant Mitigation Measures. Offers exceeding both these thresholds and market

warrant Mitigation Measures. Offers exceeding both—these thresholds and market impact thresholds and for which no sufficient explanation has been provided, shall be mitigated to the Reference Level determined as specified in Section 5.6, unless an agreement has been negotiated under the procedures set forth in *Exhibit 2*.

(a) For Supply Offers for the Real-Time Market: for intervals in which a Resource is dispatched for the purpose of relieving a transmission constraint above the level at which it otherwise would have been dispatched ("Constrained Hours"), the ISO shall assess the market impact of any Supply Offers (Section 5.5.2(b)) that meet the following thresholds:

(i) Energy Offer Price – an increase of \$25 or 50%, whichever is lower, above the Reference Level; or; or

Issued by: David T. Doot, Secretary Effective: September 15, 2002 January 21, 2003

Issued on: October January 21, 2002 2003

Economic Withholding and Uneconomic Production

- (ii) Start-Up or No-Load Price an increase of 50% above the Reference Level.
- (b) For Supply Offers for the Day-Ahead Market: for all Constrained Hours (as defined above) the ISO shall assess the market impact of any Supply Offers for the Resource that meet a threshold determined in accordance with the formula specified in subsection (a).
- **5.4** Threshold for Identifying Uneconomic Production. In addition to the thresholds governing forms of economic withholding in Sections 5.3.1, 5.3.2,5.3.2 and 5.3.3, the ISO will monitor for actions not consistent with competitive conduct involving uneconomic production. The following thresholds may warrant the imposition of a Mitigation Measure as provided in Section 5.7.4: (i) Energy scheduled at an LMP that is less than 20 % of the applicable Reference Level and causes or contributes to transmission congestion; or (ii) Real-Time output from a Resource that exceeds 110 % of the ISO's Dispatch Rate, and causes or contributes to transmission congestion.

5.5 Hourly Market Impact and Operating Reserve Thresholds.

- **5.5.1 Initial Investigation.** Before imposing any Mitigation Measure as permitted in Section 5.7, with regard to offers and bids identified in accordance with Sections 5.3.1, 5.3.2 or 5.3.3, and 5.4, the ISO shall investigate the reasons for the change in accordance with the applicable provisions of Section 3. If the offers and bids in question are not explained to the satisfaction of the ISO, the ISO, in consultation with the Independent Market Advisor, will determine whether the offers and bids in question would, if not mitigated, cause a material effect on the LMP at a Node, or clearing prices in the NEPOOL Markets or Operating Reserve charges as provided in Sections 5.5.2 and 5.5.3.
- **5.5.2 Market Impact Thresholds.** Before a Mitigation Measure is imposed on offers exceeding the conduct thresholds, the ISO will determine whether there is an impact as follows:
- (a) For offers exceeding the thresholds in Section 5.3.1, a material effect is one in excess of either of the thresholds in *Exhibit 1*, Section 1.Reserved.

Issued by: David T. Doot, Secretary Effective: September 15, 2002 January 21, 2003

Issued on: October January 21, 2002 2003

Economic Withholding and Uneconomic Production

(b) For offers exceeding thresholds in Section 5.3.2, a material effect is one in excess of the threshold specified in *Exhibit 2*, Section 2.4, or Operating Reserve payment thresholds as specified in *Exhibit 1*, Section 2. For offers exceeding thresholds in Section 5.3.3, a material effect is one in excess of the conduct threshold specified in Section 5.3.3 above or Operating Reserve payment thresholds as specified in *Exhibit 1*, Section 2.

5.5.3 Calculation of Price Impact.

- (a) When it has the capability to do so, the Market Monitoring Unit, in consultation with the Independent Market Advisor, shall determine the effect on prices or Operating Reserve payments of questioned conduct through the use of sensitivity analyses performed using the ISO's unit dispatch system software ("UDS"), unit commitment software ("UCS"), scheduling, pricing and dispatch software ("SPD") and such other computer modeling or analytic methods as the ISO, in consultation with the Independent Market Advisor, shall deem appropriate.
- (b) When a determination in accordance with paragraph (a) above is not practicable, including, but not limited to when market operations are being performed in the back-up control center during an Emergency, the ISO, in consultation with the Independent Market Advisor, shall determine the effect on prices or Operating Reserve payments of questioned conduct using the best available data and such models and methods as they shall deem appropriate. The price impact analysis will be performed to allow *ex ante* mitigation in the Day-Ahead Energy Market. *Ex ante* mitigation in the Real-Time Energy Market will be performed as soon as practicable.
- (c) The Market Monitoring Unit may, in consultation with the Independent Market Advisor, set thresholds below which it need not apply the UDS, UCS, SPD, and other systems if it is reasonable to conclude that the market impact thresholds are not likely to be violated.
- (d) In constrained areas, if appropriate models are not available as the result of limitations in hardware, software, or other technical difficulties, the ISO will manually evaluate the impact to determine if it is at least as large as the threshold value. If that is not practicable, then either of the following will be deemed to be a violation of the market impact screen for a constrained area Resource exceeding a conduct threshold specified in Section 5.3.1 or Section 5.3: (i) the scheduling of such

Issued by: David T. Doot, Secretary Effective: September 15, 2002 January 21, 2003

Issued on: July 15, 2002 January 21, 2003

Appendix A - Market Monitoring, Reporting and Market Power Mitigation

Uneconomic Production

- (b) The market impact thresholds described in Section 5.5 are exceeded.
- **5.7.3 Level of Default Offers.** A substitute mitigated offer (a "Default Offer") shall be designed to cause a Participant to offer as if it faced workable competition during a period when (i) the Participant does not face workable competition, and (ii) has responded to such condition by engaging in the physical or economic withholding. In designing and implementing Default Offers, the ISO shall seek to avoid causing a Resource to offer below its marginal cost.

5.7.4 Implementation.

- (a) The Default Offer may establish a mitigated value for one or more components of the offer for a given Resource equal to a Reference Level for that component of the Resource's offer determined as specified in Section 5.6.1.
- (b) A Resource subject to a Default Offer shall be paid the LMP or other market clearing price applicable to the output from the Resource. Accordingly, a Default Offer shall not limit the price that a Resource may receive or pay unless the Default Offer determines the LMP or other market clearing price applicable to that Resource.
- (c) Mitigation Measures shall not be applied if the price effects of the measures would cause the average day-ahead energy price in the mitigated locations or zones to rise over the entire day.
- (d) When Mitigation Measures are applied to Offer Data (as outlined in Section 5.3.1), mitigation shall be imposed from the first hour in which the impact test is met to the last hour in which the impact test is met, or for the duration of the mitigated Resource's minimum run time, whichever is longer.
- (e) The posting of the Day-Ahead schedule, rebidding period and reliability commitment run may be delayed if necessary for the completion of mitigation procedures.
- (f) Mitigation that does not affect the LMP or a clearing price in another ISO market may be applied in the settlement process.

Issued by: David T. Doot, Secretary

Issued on: January 21, 2003

EXHIBIT 1

MARKET IMPACT AND OPERATING RESERVE THRESHOLDS

1. Market Impact Threshold RESERVED

An increase of 200 % or \$100 per MW, whichever is lower, in the LMP at a Node and 200 % or \$25, whichever is lower, in any other NEPOOL market.

2. Operating Reserve Threshold OPERATING RESERVE THRESHOLD

An increase of more than 100 % in Operating Reserve Credits due to the Participant facing mitigation in a dispatch day, provided that the increase also exceeds \$10/MWh, compared to the Operating Reserve Credits calculated using Reference Levels as determined in Section 4.5 and the physical bid characteristics for the Resource. This calculation is as follows:

Where:

OR_e = Operating Reserve Energy Market Component

StartupPrice = Bid Startup Price (or Reference Level)

NoLoad Price = Bid No-Load Price (or Reference Level)

SE = Supplied Energy (or Reference Economic Minimum)

EBB = Energy Bid Block Prices (or Reference Levels)

LMP = Locational Marginal Price

Issued by: David T. Doot, Secretary Effective: September 15, 2002 January 21, 2003

Issued on: October January 21, 2002 2003

Filed to comply with order of the Federal Energy Regulatory Commission,

Docket No. ER02-2330-000,001 et al., issued September 20, 2002, 400101 FERC ¶

61,287.<u>61,344 (2002).</u>

t

 Operating Hour of the Resource associated with one continuous start-up/dispatch period when Energy was Supplied (or as determined by Reference Characteristics)

The ISO, in consultation with the Independent Market Advisor, shall determine the effect of questioned conduct on prices and Operating Reserve Charges using the best available data and such models and methods, as it deems appropriate.

Issued by: David T. Doot, Secretary Effective: September 15, 2002 January 21, 2003

Issued on: July 15, 2002 January 21, 2003

Notwithstanding the foregoing, the ISO may determine additional RMR Congestion Areas at any time on an emergency basis or as the result of new or changed circumstances; with such prior notice and opportunity for comment as is practicable under the circumstances.

1.2 Mitigation of Resources Within Designated Congestion Areas. During conditions when the ISO requires the dispatch of all capacity of all available resources within a Designated Congestion Area, sSo long as Supply Offers for Resources within the Designated Congestion Area are below the applicable Designated Congestion Area Threshold, no other congestion area Mitigation Measures will apply. Offers exceeding the applicable Designated Congestion Area Threshold that fail the market impact threshold test specified in Section 2.4 of this *Exhibit 2* will be mitigated to the applicable Reference Level as provided in *Appendix A*, Section 5.6. Notwithstanding the foregoing, Resources with Supply Offers and Reference Prices exceeding the Designated Congestion Area Threshold will only be evaluated for mitigation if they exceed the market impact threshold test specified in Section 2.4 of this *Exhibit 2*.

2. DESIGNATED CONGESTION AREA THRESHOLD DETERMINATION

2.1 Threshold Calculation. Annually, or more frequently as the ISO deems necessary, the ISO will calculate a threshold for the upcoming 12 calendar months (the "Designated Congestion Area Threshold"). Supply Offers below the threshold will not be mitigated unless they violate the thresholds applicable in all areas, *e.g.*, Section 5.3.1 of **Appendix A.** The Designated Congestion Area Threshold will be calculated as follows:

Designated Congestion Incremental Proxy Net Annual Fixed Cost
Area Threshold = CT Operating Cost + Annual Constrained Hours

Where:

Incremental Proxy = Calculated as specified in 5.6.1(B)(iii) of **Appendix A**for a combustion turbine with a 10,500 heat rate,
using a 3 month average of the 12 monthly forward
gas prices plus transportation charges

Issued by: David T. Doot, Secretary Effective: September 15, 2002 January 21, 2003

Issued on: October January 21, 2002 2003

Filed to comply with order of the Federal Energy Regulatory Commission,

Docket No. ER02-2330-000,001 et al., issued September December 20, 2002, 100 101 FERC ¶

61,287.<u>61,344 (2002).</u>

First Stage ARR Allocation

AUCTION REVENUE RIGHTS AND QUALIFIED UPGRADE AWARDS

1. INTRODUCTION

Auction Revenue Rights ("ARRs") are rights to receive FTR Auction Revenues from the sale of FTRs other than FTRs sold by FTR Holders. Qualified Upgrade Awards are rights to receive FTR Auction Revenues as provided in Section 1. ARRs shall be determined and allocated to Congestion Paying LSEs, Transmission Customers and NEMA LSEs (including any of the foregoing that are parties to Excepted Transactions that are included in the list of transactions in Attachments G and G-2 of the NEPOOL Tariff), using a four-stage process as described below (the "ARR Allocation").

2. FIRST STAGE ARR ALLOCATION

2.1 **Excepted Transactions.** In the first stage of each ARR Allocation, each entity serving load to which energy is delivered or making an External Transaction sale pursuant to an Excepted Transaction included in the list of transactions in Attachments G and G-2 of the NEPOOL Tariff, and which is the party responsible for paying Congestion Cost associated with energy purchased under the Excepted Transaction shall have the option to be allocated ARRs from the generator to the location of the load or External Node. Alternatively, each seller delivering energy pursuant to an Excepted Transaction to an entity serving load or making an External Transaction sale and which seller is the party responsible for paying Congestion Cost associated with energy purchased under the Excepted Transaction shall have the option to be allocated ARRs from the generation source to the location of the load or External Node. The lf the party responsible for paying the Congestion Cost associated with energy purchased under the Excepted Transaction which is an External Transaction will retain its existing contract rights for physical scheduling of such transaction until such party electsdoes not elect to be allocated ARRs under this Section 2. Once 2.1, then the party responsible for paying the Congestion CostARRs associated with energy purchased under the destination Node(s) of the load served by such Excepted Transaction which is an External Transaction elects to shall be allocated ARRs, the party (i) will not be able to revert back to using their contract rights for physical scheduling; and (ii) may request to be allocated ARRs, prior to each FTR Auction, either pursuant to Section 2.1 or pursuant to Section 2.2.

2.1.1 Requesting Allocation of First Stage ARRs for Excepted Transactions. In order to be eligible to receive ARRs in association with an Excepted Transaction, each entity to which energy is delivered pursuant to an Excepted Transaction or which delivers energy pursuant to an Excepted Transaction must request that it be allocated ARRs pursuant to this Section 2.1 and in accordance with the NEPOOL Manuals and ISO Administrative Procedures prior to the first stage of the ARR Allocation for an FTR Auction.

Issued by: David T. Doot, Secretary Effective: September 15, 2002 January 21, 2003

Issued on: July 15, 2002 January 21, 2003

Filed to comply with order of the Federal Energy Regulatory Commission,

Docket No. ER02-2330-001 et al., issued December 20, 2002, 101 FERC ¶ 61,344 (2002).

Substitute Original 1st Rev Sheet No. 601 Superseding Substitute Original 601

Introduction

APPENDIX E

LOAD RESPONSE PROGRAM

1. INTRODUCTION

- 1.1 Goal. The purpose of the Load Response Program ("LRP") is to facilitate load response during periods of peak electricity demand by providing appropriate incentives. Load Response Program incentives are available to any Participant or Non-Participant which, consistent with the requirements set forth herein, enrolls itself and/or one or more retail customers ("Demand Resources") to provide a reduction in their electricity consumption in the NEPOOL Control Area during peak demand periods. Non-Participants that wish to participate in the Load Response Program and have satisfied the applicable financial assurance criteria will be charged an annual service fee of \$5,000.500. The service fee will be applied to ISO expenses and may be superceded by a future provision in the ISO Tariff.
- **1.2 Eligibility.** The overall Load Response Program comprises the following individual components:

Day-Anead Demand Response Program
Real-Time 30 Minute Demand Response Program
Real-Time 2 Hour Demand Response Program
Real-Time Price Response Program
Real-Time Profiled Response Program

These programs are further defined in this Appendix E and the NEPOOL Manuals.

Demand Resources are only eligible to participate in one program at a time, except that a Demand Resource participating in the Day-Ahead Demand Response Program whose offer is not accepted in the Day-Ahead Energy Market, may participate in the Real-Time Price Response Program. Generating Resources that are already qualified as generating assets are not eligible to participate in the Load Response Program.

Any Demand Resource that is eligible to qualify as an ICAP Resource must be available, at a minimum, for the summer period (May 1 through September 30).

Issued by: David T. Doot, Secretary Effective: September 15, 2002 January 21, 2003

Issued on: October January 21, 2002 2003

Filed to comply with order of the Federal Energy Regulatory Commission,

Docket No. ER02-2330-000,001 et al., issued September December 20, 2002, 400 101 FERC ¶ 61,287.61,344 (2002).

- **1.3 Effectiveness.** The Load Response Program will be effective from the SMD Effective Date through December 31, 2004 February 28, 2006; except as provided in Section 2.1.
- 1.4 Allocation of Costs. The costs of the Real-Time Load Response Programs will be allocated to the applicable Real-Time Load Obligation on a Loadsystem Zonewide basis (commencing on the SMD Effective Date), except for the costs associated with the Internet-based Communication System (IBCS) as provided in Section 7. Until the Day-Ahead program is implemented, Real-Time program costs will be allocated to Real-Time Load Obligation on a system wide basis (calculated by summing the Real-Time Load Obligation in each Load Zone). Commencing on the date that the Day-Ahead program is implemented, the allocation of the Load Response Program costs (except for IBCS costs provided in Section 7) will change from Load Obligation on a Load Zone basis to Network Load on a Load system Zonewide basis. To the extent that a program participant's bid in the Day-Ahead Demand Response Program clears (is accepted), any charges or credits associated with such deviations will be allocated to the program participant. The balancing credit or charge will be allocated to Network Load on a Loadsystem Zonewide basis. As stated in Section 2.1, the effective date for the Day-Ahead Demand Response Program will be the date specified by the ISO and posted on its website. Such date will be at least two weeks after the ISO has given the Commission written notice and has posted on its website that the NEPOOL System Rules and computer programs necessary to implement the Day-Ahead Demand Response Program are fully in place and functional.
- **1.5 General Requirements.** A Demand Resource cannot span multiple Load Zones. A Price Response customer cannot span multiple Load Zones. All programs, except the Profiled Response Program, require interval metering. With the exception of the Profiled Response Program and "Super" Low-Tech option of the Real-Time Price Response Program, meters are read at least daily and some will require an Internet-based Communication System.

Issued by: David T. Doot, Secretary Effective: February 25, 2003

Issued on: January 21, 2003

2. DAY-AHEAD DEMAND RESPONSE PROGRAM

Demand Resources that require more than 2 hours advance notice in order to curtail consumption may participate in the Day-Ahead Demand Response Program. Demand Resources with response times less than 2 hours may also participate in the Day-Ahead Demand Response Program. The Day-Ahead Demand Response Program is not intended to pay for load reductions that would have been scheduled in any event, such as facility shut-downs.

- **2.1 Effective Date.** The Day-Ahead Demand Response Program will be effective as soon as practicable, but no earlier than the SMD Effective Date. The effective date for this program will be the date specified by the ISO and posted on its website. Such date will be at least two weeks after the ISO has given the Commission written notice that the NEPOOL System Rules and computer programs necessary to implement the Day-Ahead Demand Response Program are fully in place and functional and the ISO will post the date on its website at the time that the ISO makes such notice to the Commission.
- **2.2 Offer Parameters.** A program participant may submit Supply Offers in the Day-Ahead Energy Market on behalf of a Demand Resource in increments of 1 MW or more. Resources may be aggregated to reach the 1 MW minimum. The minimum Supply Offer shall be \$50/MWh and the maximum shall be \$500/MWh. Demand Resources that participate in the Day-Ahead Demand Response Program are eligible to qualify as an ICAP Resource subject to the performance criteria identified within the NEPOOL Manuals.
- **2.3 Payment.** Demand Resource Supply Offers that clear the Day-Ahead Energy Market will be paid the applicable Day-Ahead Zonal Price. To the extent a program participant's bid in Day-Ahead Demand Response Program clears (is accepted), and the participant's Real-Time load response deviates from its nominated response, any charges or credits associated with such deviations will be allocated to the program participant. The balancing credit or charge will be allocated to Network Load on a Load Zonesystem wide basis. Data for calculating actual performance, including the base line and actual reductions shall be provided on a daily basis with other meter reading data.

Issued by: David T. Doot, Secretary Effective: February 25, 2003

Issued on: January 21, 2003

Appendix E – Load Response Program

3. REAL-TIME DEMAND RESPONSE PROGRAMS

- **3.1 General Terms.** The Load Response Program includes two components that provide payments for Demand Resources that are willing and capable of responding to Real-Time ISO instructions to interrupt load within a specific time period. The minimum, aggregated size to participate in these programs is 100 kW.
- **3.1.1 Technical Requirements.** Both the 30-Minute Demand Response and the 2-hour Demand Response Programs require the use of an Internet-based Communication System.
- **3.1.2 Program Activation.** The ISO may issue interruption instructions to Demand Resources on a zonal or system wide (implemented in blocks) basis. The ISO may issue interruption instructions in blocks and not just by zone, to allow for a controlled implementation. A block is a system wide slice of the Demand Resources (approximately 200 MW per block).
- **3.2 30 Minute Demand Response Program.** The 30 Minute Demand Response Program requires a Demand Resource to respond within 30 minutes of the ISO's instructions to interrupt.
- **3.3** Payment **30** Minute Demand Response Program. Program participants receive the higher of the applicable Real-Time Zonal Price for interrupted consumption (measured against the base line) or a guaranteed minimum payment of \$150500/MWh for a minimum of 2 hours. Demand Resources that participate in the 30 Minute Demand Response Program are eligible to qualify as an ICAP Resource, subject to the performance criteria identified in the NEPOOL Manuals.
- **3.4 2 Hour Demand Response Program.** The 2 Hour Demand Response Program requires a Demand Resource to respond within 2 hours of the ISO's instructions to interrupt.
- 3.5 Payment 2 Hour Demand Response Program. Program participants will receive the higher of the applicable Real-Time Zonal Price for interrupted consumption (measured against the base line) or a guaranteed minimum payment of \$100350/MWh for a minimum of 2 hours. Demand Resources that participate in the 2 Hour Demand Response Program are eligible to qualify as an ICAP Resource, subject to the performance criteria identified in the NEPOOL Manuals.

Issued by: David T. Doot, Secretary Effective: January 21, 2003

Issued on: January 21, 2003

ATTACHMENT 3

ATTACHMENT 3 CLARIFICATION OF TERMS

I. Clarify the difference between Real-Time Load Obligation Deviation and Real-Time Adjusted Load Obligation Deviation.

General Information

- The definitions of these quantities can be referenced in section 3.2.1 of Market Rule 1.
- The quantities discussed below; Day-Ahead Load Obligation, Real-Time Load
 Obligation, Day-Ahead Adjusted Load Obligation, Real-Time Adjusted Load Obligation,
 Real-Time Load Obligation Deviation, and Real-Time Adjusted Load Obligation
 Deviation, are all expressed in MWh and are calculated for each Participant at each
 location (node, zone, hub) for each hour.

Real-Time Load Obligation Deviation – The Real-Time Load Obligation Deviation at a location is the algebraic difference between the Participant's Real-Time Load Obligation (RTLO) and Day-Ahead Load Obligation (DALO) at that location where:

Real-Time Load Obligation =

Revenue Metered Load (which represents the physical energy serving obligation of the Participant absent any bilateral activity)

- + Scheduled Exports
- + Internal Bilateral for Load Sales (which represent the assumption of physical energy serving obligations from another Participant)
- Internal Bilateral for Load Purchases (which represent the transfer of physical energy serving obligations to another Participant)

<u>Day-Ahead Load Obligation</u> =

Cleared Demand bids

- + Cleared Decrements Bids
- + Cleared Exports

The Real-Time Load Obligation (RTLO) is the sum of the RT physical energy serving obligations of the Participant. Combining the metered load with any external obligations and then netting any Internal Bilaterals for Load yields the RTLO. The Day-Ahead Load Obligation (DALO) is the sum of the DA cleared energy obligations of the Participant. Combining the cleared Demand and Decrement bids with Day-Ahead external obligations yields the DALO. Internal Bilaterals for Load (IBLs) are not available in the Day-Ahead market.

1 of 3 1/21/2003

Real-Time Adjusted Load Obligation Deviation – The Real-Time Adjusted Load Obligation Deviation at a location is the algebraic difference between the Participant's Real-Time Adjusted Load Obligation (RTALO) and Day-Ahead Adjusted Load Obligation (DAALO) at that location where:

<u>Real-Time Adjusted Load Obligation</u> = Real-Time Load Obligation

- + Internal Bilateral for Market (Energy) Sales (which represent the transfer of a specific quantity of energy to a Participant)
- Internal Bilateral for Market (Energy) Purchases (which represent the acquisition of a specific quantity of energy from a Participant)

<u>Day-Ahead Adjusted Load Obligation</u> = Day-Ahead Load Obligation

- + Internal Bilateral for Market (Energy) Sales (which represent the transfer of a specific quantity of energy to a Participant)
- Internal Bilateral for Market (Energy) Purchases (which represent the acquisition of a specific quantity of energy from a Participant)

By looking at the equations, we can see the only difference between the Real-Time Load Obligation and Real-Time Adjusted Load Obligation terms is the inclusion of Internal Bilateral for Market (Energy) transactions

II. Distinguish between Internal Bilateral for Load and Internal Bilateral for Market (Energy)

Internal Bilateral for Load (IBL) – For the purposes of Energy Market accounting, An IBL will increase or decrease the energy serving obligations of the two Participants involved in the transaction. IBLs do not impact the physical dispatch of the power system and are reflected only in the settlement process. An IBL can only be used in the Real-Time Market. In addition to their impact on Energy Market positions, IBLs impact the Regulation market positions of both parties and, under current rules, shift responsibility for Operating Reserve charges from the buyer to the seller.

Internal Bilateral for Market (IBM) – An IBM transfers the rights to a specific quantity of a specified Market product between two Participants. They are equivalent to system contracts in New England's existing market and are not associated with a specific resource. IBMs do not impact the physical dispatch of the power system and are reflected only in the settlement process. An IBM will only affect the market for which the transaction was submitted. An IBM for Energy will only affect the Participant's position in the Energy Market and not the Regulation or ICAP Market. A Participant's may submit three separate transactions, one to cover each Market,

2 of 3 1/21/2003

if it wants to cover all products. In the Day-Ahead Market only the IBM for Energy can be used. An IBM for Regulation is only available in the Real-Time market and the ICAP IBM can only be used for ICAP. IBMs currently have no impact on the Operating Reserve charge allocation.

The table below indicates the markets impacted by each type of bilateral transaction. Note that Day-Ahead IBMs for Energy are shown as having an impact in the Real-Time markets. This is because, under the current rules, a Day-Ahead IBM for Energy is automatically carried forward as a Real-Time IBM for Energy.

Type	Product	DA	RT	RT	RT	Monthly
		Energy	Energy	Regulation	Operating	ICAP
					Reserve	
					Charges	
IBM	Energy	X	X			
	(DA)					
IBM	Energy		X			
	(RT)					
IBM	Regulation			X		
IBM	ICAP					X
IBL			X	X	X	

3 of 3 1/21/2003

ATTACHMENT 4

NEPOOL PARTICIPANTS COMMITTEE VOTING RESULTS January 10, 2003

<u>SECTOR</u>	VOTE 1	VOTE 2	VOTE 3
GENERATION	0.00	15.00	16.00
TRANSMISSION	8.57	20.00	0.00
SUPPLIER	8.57	15.56	18.57
PUBLICLY OWNED ENTITY	20.00	20.00	10.00
END USER	4.29	20.00	0.00
% IN FAVOR	41.43	90.56	44.57
	41.43	90.56	44.57
	100.00 0.00	100.00 0.00	100.00 0.00
	0.00	0.00	0.00

GENERATION SECTOR

Participant Name	Vote 1	Vote 2	Vote 3
1 ANP Marketing			
1 Duke Energy North America LLC			
1 Entergy Nuclear Generation Company	0	А	F
1 FPL Energy LLC	0	F	0
1 Generation Group Member	0	0	F
1 Mirant New England, LLC	0	F	F
1 TransCanada Power Marketing Ltd.	Α	F	F
7			
IN FAVOR (F)	0	3	4
OPPOSED (O)	4	1	1
TOTAL VOTES	4	4	5
ABSTENTIONS (A)	1	1	0
ALL VOTES			
ALL VOTES	5 Y	5 Y	5 Y
QUORUM IN ATTENDANCE	•	Y	Y
SECTOR QUORUM	3		
MEMBER FIXED VOTING SHARE	2.857		
UNADJUSTED SECTOR VOTING SHARE	0	15	16
ADJUSTED SECTOR VOTING SHARE	0	15	16
SVS/SUM MFVS	20	20	20

TRANSMISSION SECTOR

	Participant Name	Vote 1	Vote 2	Vote 3
1	Bangor Hydro-Electric Company	0	F	
1	Boston Edison Company	0	F	0
1	Central Maine Power Company	F	F	0
1	New England Power Company	0	F	0
1	Northeast Utilities Service Company	F	Α	0
1	The United Illuminating Company	0	F	0
1	Vermont Electric Power Company, Inc.	F	F	0
7				
	IN FAVOR (F)	3	6	0
	OPPOSED (O)	4	0	6
	TOTAL VOTES	7	6	6
	ABSTENTIONS (A)	0	1	0
	ALL VOTES	7	7	6
	QUORUM IN ATTENDANCE	Υ	Υ	Υ
	SECTOR QUORUM	3		
	MEMBER FIXED VOTING SHARE	2.857		
	SECTOR VOTING SHARE	8.57	20.00	0.00
	ADJUSTED SECTOR VOTING SHARE	8.571428571	20	0
	SVS/SUM MFVS	20.00	20.00	20.00

	Participant Name	Vote 1	Vote 2	Vote 3
1	AES Londonderry	Α	F	F
1	AIG Energy Trading Inc.			
1	Allegheny Energy Supply Company			
	Allied Utility Network LLC			
1	American Electric Power Service Corp.			
1	Aquila Merchant Services			
	BP Energy Company			
	Brascan Energy Marketing Inc.			
1	Calpine Energy Services, L.P.	A	F	F
	Cargill Power Markets, LLC			
	CinCap V, LLC			
	Citadel Energy Products LLC			
	Conectiv Energy Supply, Inc.			
	Consolidated Edison Energy, Inc.	Α	F	
	Constellation Power Source, Inc.	0	F	0
	Coral Power, LLC	A	A	F
	DTE Energy Trading, Inc.			
	Dynegy (f/k/a Electric Clearing House, Inc.)			
	Edison Mission Marketing and Trading			
	El Paso Merchant Energy, LP	0	Α	F
	Energy America, LLC		, \	
	Energy Atlantic, LLC			
	Exelon Generation Company, LLC	Α	Α	F
	Great Bay Power Marketing, Inc.		Λ	I I
	Hess Energy Power & Gas Company, LLC			
	H.Q. Energy Services (U.S.) Inc.	0	^	F
	Indeck-Pepperell Power Associates, Inc.	A	A A	F
		Α	Α	Г
	J. Aron & Company LIPA	Λ	^	
		A	Α	F
	Morgan Stanley Capital Group, Inc.	Δ	Δ	
	NRG Power Marketing, Inc.	A	A	F
	PG&E Energy Trading	F	A	F
	PPL EnergyPlus, LLC			
	Providence Energy Services, Inc.			
	PSEG Energy Resources & Trade LLC	Α	F	F
	Reliant Energy Services, Inc.			
	RWE Trading Americas Inc.			
	Sempra Energy Trading Corp.	Α	F	A
	SmartEnergy.com, Inc.			
	Sprague Energy Corp.			
	Strategic Energy Ltd.	F	0	Α
	Tractebel Energy Marketing, Inc.			
	TransAlta Energy Marketing (U.S.) Inc.			
	TXU Energy Trading Company LP			
	UAE Lowell Power LLC	Α	A	F
	UBS AG			
	Unitil Corporation Participant Companies	0	F	
1	Virginia Electric and Power Company	F	0	F
	Williams Energy Marketing & Trading Co.			
1	WPS Energy Services Inc.			
50		-		
	IN FAVOR (F)	3	7	13
	OPPOSED (O)	4	2	1
	TOTAL VOTES	7	9	14
	ABSTENTIONS (A)	11	9	2
	· ·		-	
413	89926_1.XLS ALL VOTES	18	18	16
1/21				

	Participant Name	Vote 1	Vote 2	Vote 3
1	Ashburnham Municipal Light Plant	F	F	F
1	Belmont Municipal Light Department	F	F	F
1	Boylston Municipal Light Department	F	F	F
1	Braintree Electric Light Department	F	F	F
1	Chicopee Municipal Lighting Plant	F	F	F
1	Concord Municipal Light Plant	F	F	F
1	Conn. Municipal Electric Energy Cooperative	F	F	F
1	Danvers Electric Division	F	F	F
1	Georgetown Municipal Light Department	F	F	F
1	Groton Electric Light Department	F	F	F
1	Hingham Municipal Lighting Plant	F	F	F
1	Holden Municipal Light Department	F	F	F
1	Holyoke Gas & Electric Department	F	F	F
1	Hudson Light and Power Department	F	F	F
1	Hull Municipal Lighting Plant	F	F	F
1	Ipswich Municipal Light Department	F	F	F
1	Littleton Electric Light Department	F	F	F
1	Mansfield Municipal Electric Department	F	F	F
1	Marblehead Municipal Light Department	F	F	F
1	Mass. Municipal Wholesale Electric Company	F	F	F
1	Middleborough Gas and Electric Department	F	F	F
1	Middleton Municipal Electric Department	F	F	F
1	New Hampshire Electric Cooperative, Inc.	F	F	F
1	North Attleborough Electric Department	F	F	F
1	Norwood Municipal Light Department	-	-	-
1	Pascoag Utility District	F	F	F
1	Paxton Municipal Light Department	F	F	F
1	Peabody Municipal Light Plant	F	F	F
1	Reading Municipal Light Department			-
1	Rowley Municipal Lighting Plant	F	F	F
1	Shrewsbury's Electric Light Plant	F	F	F
1	South Hadley Electric Light Department	F	F	F
1	Sterling Municipal Electric Light Department	F	F	F
1	Taunton Municipal Lighting Plant	F	F	F
1	Templeton Municipal Lighting Plant	F	F	F
1	Vermont Public Power Supply Authority	F	F	F
1	Wakefield Municipal Gas and Light Department	F	F	F
1	Wellesley Municipal Light Plant	'		'
1	West Boylston Municipal Lighting Plant	F	F	F
1	Westfield Gas & Electric Light Department	F	F	F
40	Treation Cao a Elouno Eight Dopartinont	•	'	
	IN FAVOR (F)	37	37	37
	OPPOSED (O)	0	0	37
	TOTAL VOTES	37	37	74
	ABSTENTIONS (A)	0	0	0
	ADOTENTIONS (A)	J	0	J
	ALL VOTES	37	37	74
	QUORUM IN ATTENDANCE	Y	Y	Y
	SECTOR QUORUM	3		

END USER SECTOR

	Participant Name	Vote 1	Vote 2	Vote 3
1	Associated Industries of Massachusetts (O)	0	F	0
	Cape Light Compact, The (O)			
	Connecticut, State of, Office of Consumer Counsel (O)	0	F	0
1	Energy Management Inc. (S)			
1	Forster, Inc. (S)	0	F	0
1	Gardiner Paperboard (L)	0	F	0
1	Industrial Energy Consumer Group (O)	0	F	0
	J.F. Gray & Associates, LLC (S)			
1	LaPorta, Leonard (S)			
1	Maine Health & Higher Educ. Fac. Authority, The (O)			
	Maine Skiing, Inc. (O)	0	F	0
1	Maine, State of, Office of the Governor (O)	Α	F	Α
	Mass. Energy Buyers Coalition (MEBC) (O)			
1	Mass. Public Interest Research Group (MASSPIRG) (O)	F		
1	McLaughlin Trust, Robert E. (S)			
	Mead Oxford Corporation (S)	0	F	0
	New Hampshire Office of Consumer Advocate (O)	F	A*	0
	PowerOptions, Inc. (O)			
1	Praxair, Inc. (L)	0	F	0
	Schaefer, Marc (S)			
1	Silkman, Richard (S)			
1	Texas Instruments (L)	0	F	0
1	The Energy Consortium (O)	0	F	0
1		0	F	0
1	Union of Concerned Scientists (O)	F	F	0
1	WebGen Systems, Inc. (S)			
26	, , ,			
	*NHOCA subsequently requested to abstain which does	not effect the fir	nal results.	
	IN FAVOR (F)	3	13	0
	OPPOSED (O)	11	0	13
	TOTAL VOTES	14	13	13
	ABSTENTIONS (A)	1	1	1
	ALL VOTES	15	14	14
	QUORUM IN ATTENDANCE	Υ	Υ	Υ
	SECTOR QUORUM	3		
	MEMBER FIXED VOTING SHARE	0.769		
	SECTOR VOTING SHARE	4.29	20.00	0.00
	ADJUSTED SECTOR VOTING SHARE	4.285714286	20	0
	SVS/SUM MFVS	20.00	20.00	20.00

ATTACHMENT 5

NEPOOL Participants Committee Members and Alternates

Chuck Sjoberg

AES Londonderry

c/o AES Odyssey

1001 West Seneca Street

Suite 100

Ithaca, NY 14850

Chris Wentlent (Alt)

AES Londonderry

c/o AES Westover

720 Riverside Drive

Johnson City, NY 13790

Carl Peterson

AIG Energy Trading, Inc.

One Greenwich Plaza

Greenwich, CT 06830

Hilary Garfein (Alt)

AIG Energy Trading, Inc.

One Greenwich Plaza

Greenwich, CT 06830

Joseph Limone

Director & Senior Counsel

Allegheny Energy Supply Company, LLC

909 Third Avenue, 33rd Floor

New York, NY 10022

Laurie K. Peddicord (Alt)

Contract Administrator

Allegheny Energy Supply Company, LLC

909 Third Avenue, 33rd Floor

New York, NY 10022

Michele Brown

Vice President

Allied Utility Network

3346 Nottingham Forest Drive

Gainesville, GA 30506

Michael Horne (Alt)

Vice President – Business Development

Allied Utility Network

3910 Gallant Fox CT.

Duluth, GA 30096

January 21, 2003

Ron McNamara

American Electric Power Service Corporation

One Riverside Plaza

14th Floor

Columbus, OH 43215

Thomas Chapman (Alt)

American Electric Power Service Corporation

One Riverside Plaza

14th Floor

Columbus, OH 43215

Dorothy J. Capra

Director Regulatory Affairs

ANP Marketing Company

65 Boston Post Road

Suite 300

Marlborough, MA 01752

William Henson (Alt)

Director, Power Marketing

ANP Marketing Company

65 Boston Post Road

Suite 300

Marlborough, MA 01752

Kevin J. Fox

Portfolio Manager & Vice President

Aquila Merchant Services, Inc.

1100 Walnut, Suite 3300

Kansas City, MO 64106

Peter W. Brown (Alt)

Attorney

Aquila Merchant Services, Inc.

c/o Brown, Olson & Wilson, P.C.

501 South Street

Concord, NH 03304

Stanley W. Herriott

Manager

Ashburnham Municipal Light Plant

78 Central Street

P.O. Box 823

Ashburnham, MA 01430

NEPOOL Participants Committee Members and Alternates

Roger W. Bacon (Alt)

Ashburnham Municipal Light Plant

c/o Mass. Municipal Wholesale Electric Co.

Moody Street

P.O. Box 426

Ludlow, MA 01056-0246

Robert Ruddock

Executive Vice President

Associated Industries of Massachusetts

Suite 1300

222 Berkeley Street

Boston, MA 02117

Angela M. O'Connor (Alt)

Vice President of Energy Programs

Associated Industries of Massachusetts

13th Floor

222 Berkeley Street

Boston, MA 02117

Jeffrey A. Jones

Manager-Legal & Regulatory Services

Bangor Hydro-Electric Company

33 State Street

P.O. Box 932

Bangor, ME 04402-0932

Gregory Hines (Alt)

Director of Business Services

Bangor Hydro-Electric Company

33 State Street

P.O. Box 932

Bangor, ME 04402-0932

Timothy L. McCarthy

Manager

Belmont Municipal Light Department

450 Concord Avenue

Belmont, MA 02478

Roger W. Bacon (Alt)

Belmont Municipal Light Department

c/o Mass. Municipal Wholesale Electric Co.

Moody Street

P.O. Box 426

Ludlow, MA 01056-0246

January 21, 2003

Ellen K. Angley

Vice President, Energy Supply and

Transmission

Boston Edison Company

c/o NSTAR

800 Boylston Street, P1603

Boston, MA 02199-8001

Robert P. Clarke (Alt)

Director, Energy Planning and Supply

Boston Edison Company

c/o NSTAR

800 Boylston Street (P1704)

Boston, MA 02199-8001

H. Bradford White, Jr.

Manager

Boylston Municipal Light Department

Paul X. Tivnan Road

P.O. Box 753

Boylston, MA 01505-0753

Roger W. Bacon (Alt)

Boylston Municipal Light Department

c/o Mass. Municipal Wholesale Electric Co.

Moody Street

P.O. Box 426

Ludlow, MA 01056-0246

Robert Scavone

BP Energy Company

501 WestLake Park Boulevard

Houston, TX 77079

Jason Stever (Alt)

BP Energy Company

501 WestLake Park Boulevard

Houston, TX 77079

Laurie J. Heffron

Electric Operations Manager

Braintree Electric Light Department

150 Potter Road

Braintree, MA 02184-3598

NEPOOL Participants Committee Members and Alternates

John P. Coyle (Alt)

Braintree Electric Light Department

c/o Duncan & Allen 1575 Eye Street, NW Washington, DC 20005-1175

Tom Kaslow

Calpine Energy Services, LP

c/o Calpine Eastern Corporation The Pilot House, 2nd Floor Lewis Wharf Boston, MA 02110

Jeff Woodall (Alt) Lead Scheduler

Calpine Energy Services, LP

700 Louisiana Ave., Ste. 2700

Houston, TX 77002

Margaret Downey

Asst. County Administrator

Cape Light Compact, The

3195 Main Street P.O. Box 427

Barnstable, MA 02630

Robert Bigelow (Alt)

Cape Light Compact, The

P.O. Box 165

Woods Hole, MA 02543

John Trimble

Business Development Manager

Cargill Power Markets, LLC

12700 Whitewater Drive

Minnetonka, MN 55343-9439

Gaston Garrido (Alt)

Trading Manager

Cargill Power Markets, LLC

12700 Whitewater Drive

Minnetonka, MN 55343-9439

Eric N. Stinneford

Manager, Power Contracts Administration

Central Maine Power Company

41 Anthony Avenue

Augusta, ME 04330

January 21, 2003

Hariph M. Smith (Alt)

Central Maine Power Company

41 Anthony Avenue Augusta, ME 04330

Jeffrey R. Cady

Power Resource Manager

Chicopee Municipal Lighting Plant

725 Front Street

P.O. Box 405

Chicopee, MA 01021-0405

Brian Forshaw

Chicopee Municipal Lighting Plant

c/o Connecticut Municipal Electric Energy

Cooperative

30 Stott Avenue

Norwich, CT 06360-1535

Jason Barker

The Cincinnati Gas & Electric Company, Inc.

139 East 4th Street

EA503

Cincinatti, OH 45202

Walt Yaeger (Alt)

The Cincinnati Gas & Electric Company, Inc.

139 East 4th Street

EA503

Cincinatti, OH 45202

Murat Alptekin

Director

Citadel Energy Products LLC

c/o Citadel Investment Group, LLC 225 W. Washington St., 9th Fl.

Chicago, IL 60606

Peter Sullivan (Alt)

Vice President

Citadel Energy Products LLC

c/o Citadel Investment Group, LLC 225 W. Washington St., 9th Fl.

Chicago, IL 60606

Daniel J. Sack Superintendent

Concord Municipal Light Plant

1175 Elm Street P.O. Box 1029

Concord, MA 01742-1029

Roger W. Bacon (Alt)

Concord Municipal Light Plant

c/o Mass. Municipal Wholesale Electric Co.

Moody Street P.O. Box 426

Ludlow, MA 01056-0246

Marj Garbini

Transmission Engineer

Conectiv Energy Supply, Inc.

P.O. Box 6066

Newark, DE 19714-6066

Bill Fehr (Alt)

Power Trading Manager

Conectiv Energy Supply, Inc.

P.O. Box 6066

Newark, DE 19714-6066

Brian E. Forshaw

Director

Connecticut Municipal Electric Energy Cooperative

30 Stott Avenue

30 Stott Avenue

Norwich, CT 06360-1535

Maurice R. Scully (Alt)

Executive Director

Connecticut Municipal Electric Energy

Cooperative

30 Stott Avenue

Norwich, CT 06360-1535

Ken Bekman

Director, Wholesale Power

Consolidated Edison Energy, Inc.

701 Westchester Avenue

Suite 320E

White Plains, NY 10604

January 21, 2003

Michael Forte (Alt) General Manager

Consolidated Edison Energy, Inc.

4 Irving Place, Room 1349S

New York, NY 10003

Harvey J. Reed

Managing Director

Constellation Power Source, Inc.

111 Market Place

Suite 500

Baltimore, MD 21202

Daniel W. Allegretti (Alt)

Constellation Power Source, Inc.

One Essex Drive

Bow, NH 03304

Matthew Picardi

Coral Power, LLC

General Manager - Regulatory Affairs

27 Winthrop Rd.

Hingham, MA 02043

Robert Reilley (Alt)

Vice President-Regulatory Affairs

Coral Power, LLC

909 Fannin Street

Suite 700

Houston, TX 77010

Coleen O'Brien-Pitts

Electric Utility Director

Danvers Electric Division

One Burroughs Street

Danvers, MA 01923-2702

Francis M. Gaffney (Alt)

Power Supply Engineer

Danvers Electric Division

One Burroughs Street

Danvers, MA 01923-2702

Ray Wieszczyk Manager of Operations

DTE Energy Trading, Inc.

101 N. Main Street

Suite 300

Ann Arbor, MI 48104

Anthony Wisely (Alt)

Retail Trader

DTE Energy Trading, Inc.

101 N. Main Street

Suite 300

Ann Arbor, MI 48104

Stacy L. Dimou

Duke Energy North America LLC

P.O. Box 426

Millinocket, ME 04462

George Johnson (Alt)

Attorney

Duke Energy North America LLC

c/o Dickstein, Shapiro, Morin & Oshinsky LLP 2101 L Street, N.W.

Washington, DC 20037-1526

Steve Vavrik Director

Dynegy Marketing & Trade 1000 Louisiana Suite 5800

Houston, TX 77002

Miles Allen (Alt) Sr. Vice President

Dynegy Power Marketing, Inc.

101 Merrimic Street, 2nd Floor

Boston, MA 02114

William Roberts

SVP Power Contracting

Edison Mission Marketing & Trading, Inc.

160 Federal Street

Boston, MA 02110-1776

January 21, 2003

Jeffry Ellis (Alt)

Edison Mission Marketing & Trading, Inc.

160 Federal Street

Boston, MA 02110-1776

Tim Cotnet

El Paso Merchant Energy, LP

70 Walnut Street

Wellesley Hills, MA 02481

Robert de R. Stein (Alt) Principal Consultant

El Paso Merchant Energy, LP

c/o Signal Hill Consulting Group

128 Merchants Row

Suite 703

Rutland, VT 05701

Calvin P. Deschene

Director

Energy Atlantic, LLC

830 Main Street

Suite 20

P.O. Box 1148

Presque Isle, ME 04769-1148

Don Theriault (Alt)

Forecast Analyst

Energy Atlantic, LLC

P.O. Box 1148

Presque Isle, ME 04769-1148

Dennis J. Duffy

Energy Management, Inc.

One Energy Road

North Dartmouth, MA 02747

Chris Sherman (Alt)

Energy Management, Inc.

One Energy Road

North Dartmouth, MA 02747

Alan Cherkas

Executive Director, Structured Power, NE

Engage Energy America LLC

39500 High Pointe Blvd.

Suite 260

Novi, MI 48375

Terrence O'Reilly (Alt)
Vice President & Senior Counsel

Engage Energy America LLC

39500 High Pointe Blvd.

Suite 260

Novi, MI 48375

Brent Dorsey

Entergy Nuclear Generation Company

P.O. Box 3200 Jackson, MS 39286

William Stone (Alt)

Entergy Nuclear Generation Company

600 Rocky Hill Road Plymouth, MA 03260

Mike Roan

Exelon Generation Company, LLC

c/o Exelon New England Holdings, LLC

The Schrafft Center 529 Main Street Suite 605

Charlestown, MA 02129

Jack Crowley (Alt)

Exelon Generation Company, LLC

c/o Exelon New England Holdings, LLC

The Schrafft Center 529 Main Street Suite 605

Charlestown, MA 02129

Donald J. Sipe

Forster, Inc.

c/o Preti, Flaherty, Beliveau, Pachios & Haley 45 Memorial Circle

P.O. Box 1058

Augusta, ME 04332-1058

Benjamin Bilus (Alt)

Forster, Inc.

c/o Preti, Flaherty, Beliveau, Pachios & Haley

45 Memorial Circle

P.O. Box 1058

Augusta, ME 04332-1058

January 21, 2003

Roberto R. Denis

Vice President

FPL Energy LLC

9250 W. Flagler Street

Suite 5494

Miami, FL 33174

Fernando DaSilva (Alt)

Director, Market Affairs

FPL Energy LLC

8 Woodlawn Road

Assonet, MA 02702

Donald J. Sipe

Gardiner Paperboard

c/o Preti, Flaherty, Beliveau, Pachios & Haley

45 Memorial Circle

P.O. Box 1058

Augusta, ME 04332-1058

Benjamin Bilus (Alt)

Gardiner Paperboard

c/o Preti, Flaherty, Beliveau, Pachios & Haley

45 Memorial Circle

P.O. Box 1058

Augusta, ME 04332-1058

William P. Short III

Generation Group Member

c/o Ridgewood Power Management, LLC

947 Linwood Avenue

Ridgewood, NJ 07450

William Rodgers

Director of Marketing

Great Bay Power Marketing, Inc.

51 Dow Highway, Suite 7

Eliot, ME 03903

Wayne Snow

Manager

Georgetown Municipal Light Department

94 Searle Street

Georgetown, MA 01833

Roger W. Bacon (Alt)

Georgetown Municipal Light Department

c/o Mass. Municipal Wholesale Electric Co.

Moody Street

P.O. Box 426

Ludlow, MA 01056-0246

Doris Chojowski

Manager

Groton Electric Light Department

23 Station Avenue

Groton, MA 01450-4222

Roger W. Bacon (Alt)

Groton Electric Light Department

c/o Mass. Municipal Wholesale Electric Co.

Moody Street

P.O. Box 426

Ludlow, MA 01056-0246

Mark Hallenbeck

Hess Energy Power & Gas Company, LLC

1185 Avenue of the Americas

New York, NY 10036

Charles F. Cerria (Alt)

Hess Energy Power & Gas Company, LLC

1185 Avenue of the Americas

New York, NY 10036

Christian Desilets

H.Q. Energy Services (U.S.) Inc.

75, boul. Rene-Levesque Ouest

17e etage

Montreal, Quebec H2Z 1A4

CANADA

Robert Stein (Alt)

Principal Consultant

H.Q. Energy Services (U.S.) Inc.

c/o Signal Hill Consulting Group

128 Merchants Row

Suite 703

Rutland, VT 05701

January 21, 2003

Joseph R. Spadea Jr.

Manager

Hingham Municipal Lighting Plant

222 Central Street

Hingham, MA 02043-2518

Roger W. Bacon (Alt)

Hingham Municipal Lighting Plant

c/o Mass. Municipal Wholesale Electric Co.

Moody Street

P.O. Box 426

Ludlow, MA 01056-0246

Edla A. Bloom

Director

Holden Municipal Light Department

94 Reservoir Street

Holden, MA 01520

Roger W. Bacon (Alt)

Holden Municipal Light Department

c/o Mass. Municipal Wholesale Electric Co.

Moody Street

P.O. Box 426

Ludlow, MA 01056-0246

James M. Lavelle

Manager

Holyoke Gas & Electric Department

99 Suffolk Street

Holyoke, MA 01040

Brian C. Beauregard (Alt)

Superintendent

Holyoke Gas & Electric Department

99 Suffold Street

Holyoke, MA 01040

Anthony J. Monteiro

General Manager

Hudson Light & Power Department

49 Forest Avenue

Hudson, MA 01749

Roger W. Bacon (Alt)

Hudson Light & Power Department

c/o Mass. Municipal Wholesale Electric Co.

Moody Street

P.O. Box 426

Ludlow, MA 01056-0246

John A. MacLeod

Manager

Hull Municipal Lighting Plant

15 Edgewater Road Hull, MA 02045-2714

Roger W. Bacon (Alt)

Hull Municipal Lighting Plant

c/o Mass. Municipal Wholesale Electric Co.

Moody Street

P.O. Box 426

Ludlow, MA 01056-0246

Maurice T. Klefeker

Asset Manager

Indeck Pepperell Power Associates, Inc.

1075 Noel Avenue

Wheeling, IL 60090

Mark V. Magyar (Alt)

Consultant

Indeck Pepperell Power Associates, Inc.

c/o Financial Management Co.

325A Southbridge Street

Auburn, MA 01501

Donald J. Sipe

Industrial Energy Consumer Group

c/o Preti, Flaherty, Beliveau, Pachios & Haley

45 Memorial Circle

P.O. Box 1058

Augusta, ME 04332-1058

Benjamin Bilus (Alt)

Industrial Energy Consumer Group

c/o Preti, Flaherty, Beliveau, Pachios & Haley

45 Memorial Circle

P.O. Box 1058

Augusta, ME 04332-1058

January 21, 2003

Raymond R. Shockey

Manager

Ipswich Municipal Light Department

P.O. Box 151

272 High Street

Ipswich, MA 01938-0151

Roger W. Bacon (Alt)

Ipswich Municipal Light Department

c/o Mass. Municipal Wholesale Electric Co.

Moody Street

P.O. Box 426

Ludlow, MA 01056-0246

Gordon van Welie

ISO New England Inc.

One Sullivan Road

Holyoke, MA 01040-2841

Kevin Kirby (Alt)

ISO New England Inc.

One Sullivan Road

Holyoke, MA 01040-2841

Richard Ruzika

Managing Director

J. Aron & Company

85 Broad Street

New York, NY 10004

Scott Weinstein (Alt)

Managing Director

J. Aron & Company

85 Broad Street

New York, NY 10004

Robert J. Gray

President

J.F. Grav & Associates LLC

35 Woodman Road

South Hampton, NH 03827

John F. Gray (Alt)

J.F. Gray & Associates LLC

35 Woodman Road

South Hampton, NH 03827

LaPorta, Leonard

150 West Newton Street Boston, MA 02118

Jim Parmelee

Director of Power Markets

LIPA

31 Brinker Drive South Rensselaer, NY 12144

John Higgins (Alt)

LIPA

c/o Navigant Consulting Inc. 200 Wheeler Road, Suite 4 Burlington, MA 01803

Scott Edwards

Assistant General Manager

Littleton Electric Light & Water Department

39 Ayer Road P.O. Box 2406

Littleton, MA 01460-3406

Roger W. Bacon (Alt)

Littleton Electric Light & Water Department

c/o Mass. Municipal Wholesale Electric Co.

Moody Street

P.O. Box 426

Ludlow, MA 01056-0246

Paul Jutras

Maclaren Energy Inc.

2 Montreal Road West

Masson-Angers, Quebec, CANADA J8M 1K6

Richard St. Jean (Alt)

Maclaren Energy Inc.

2 Montreal Road West

Masson-Angers, Quebec, CANADA J8M 1K6

Stephen M. Gauthier

Program Manager

Maine Health & Higher Educational

Facilities Authority

3 University Drive

P.O. Box 2268

Augusta, ME 04338-2268

January 21, 2003

Robert O. Lenna (Alt)

Executive Director

Maine Health & Higher Educational

Facilities Authority

3 University Drive

P.O. Box 2268

Augusta, ME 04338-2268

Donald J. Sipe

Maine Skiing, Inc.

c/o Preti, Flaherty, Beliveau, Pachios & Haley

45 Memorial Circle

P.O. Box 1058

Augusta, ME 04332-1058

Benjamin Bilus (Alt)

Maine Skiing, Inc.

c/o Preti, Flaherty, Beliveau, Pachios & Haley

45 Memorial Circle

P.O. Box 1058

Augusta, ME 04332-1058

John J. Beliveau

Director

Mansfield Electric Light Department

125 High Street, Suite 4

Mansfield, MA 02048-2404

Roger W. Bacon (Alt)

Mansfield Electric Light Department

c/o Mass. Municipal Wholesale Electric Co.

Moody Street

P.O. Box 426

Ludlow, MA 01056-0246

Robert V. Jolly Jr.

Manager

Marblehead Municipal Light Department

80 Commercial Street

P.O. Box 369

Marblehead, MA 01945-0369

Roger W. Bacon (Alt)

Marblehead Municipal Light Department

c/o Mass. Municipal Wholesale Electric Co.

Moody Street

P.O. Box 426

Ludlow, MA 01056-0246

Neil Chayet

Massachusetts Energy Buyers Coalition

ML Strategies/MEBC c/o Jennifer White One Financial Center Boston, MA 02111

Douglas Stevenson (Alt)

Massachusetts Energy Buyers Coalition

ML Strategies/MEBC One Financial Center Boston, MA 02111

Roger W. Bacon

Director, Power Services Division

Mass. Municipal Wholesale Electric Co.

Moody Street P.O. Box 426

Ludlow, MA 01056-0246

Timothy Peet (Alt)

Mass. Municipal Wholesale Electric Co.

Moody Street P.O. Box 426

Ludlow, MA 01056-0246

Frank Gorke

Massachusetts Public Interest Research Group

29 Temple Place 5th Floor

Boston, MA 02111

Robert Sargent (Alt)

Massachusetts Public Interest Research Group

29 Temple Place 5th Floor Boston, MA 02111

Donald J. Sipe

Mead Oxford Corporation

c/o Preti, Flaherty, Beliveau, Pachios & Haley 45 Memorial Circle P.O. Box 1058

Augusta, ME 04332-1058

January 21, 2003

Benjamin Bilus (Alt)

Mead Oxford Corporation

c/o Preti, Flaherty, Beliveau, Pachios & Haley

45 Memorial Circle P.O. Box 1058

Augusta, ME 04332-1058

James L. Collins Energy Analyst

Middleborough Gas & Electric

2 Vine Street

Middleborough, MA 02346

Mayhew D. Seavey (Alt)

Principal

Middleborough Gas & Electric

c/o PLM, Inc. 35 Main Street

Hopkinton, MA 01748

Mark T. Kelly Manager

Middleton Municipal Light Department

197 North Main Street Middleton, MA 01949-1068

Roger W. Bacon (Alt)

Middleton Municipal Light Department

c/o Mass. Municipal Wholesale Electric Co.

Moody Street P.O. Box 426

Ludlow, MA 01056-0246

Peter D. Fuller

Mirant New England, LLC

1099 Hingham Street Rockland, MA 02370

Michael R. Kirkwood (Alt)

Mirant New England, LLC

1099 Hingham Street Rockland, MA 02370

Catherine Flax Senior Trader

Morgan Stanley Capital Group Inc.

1585 Broadway 4th floor

New York, NY 10036

Karen Kochonies (Alt)

Vice President, Gas & Electric Operations

Morgan Stanley Capital Group Inc.

1585 Broadway 4th Floor

New York, NY 10036

Michael Hager

New England Power Company

25 Research Drive

Westborough, MA 01582

Mary Ellen Paravalos (Alt)

New England Power Company

25 Research Drive

Westborough, MA 01582

Fred C. Anderson General Manager

New Hampshire Electric Cooperative, Inc.

579 Tenney Mountain Highway Plymouth, NH 03264-3147

Stephen Kaminski (Alt)

Director, Energy Access & Services

New Hampshire Electric Cooperative, Inc.

579 Tenney Mountain Highway Plymouth, NH 03264-3147

Michael W. Holmes Consumer Advocate

New Hampshire Office of Consumer Advocate

117 Manchester Street Concord, NH 03301

Paul Peterson (Alt)

New Hampshire Office of Consumer Advocate

c/o Synapse Energy Economics, Inc.

22 Pearl Street

Cambridge, MA 02139

January 21, 2003

Mark V. Magyar

North Attleborough Electric

c/o Financial Management Group

325A Southbridge Street

Auburn, MA 01501

Roger W. Bacon (Alt)

North Attleborough Electric

c/o Mass. Municipal Wholesale Electric Co.

Moody Street

P.O. Box 426

Ludlow, MA 01056-0246

Frank P. Sabatino

Senior Vice President - Power Marketing

Northeast Utilities System Companies

c/o Select Energy, Inc.

107 Selden Street

Berlin, CT 06037

James R. Shuckerow (Alt)

Director-Wholesale Power Contracts

Northeast Utilities System Companies

P.O. Box 270

Hartford, CT 06141-0270

Malcolm N. McDonald

Superintendent

Norwood Municipal Light Department

206 Central Street

Norwood, MA 02062-3567

Joseph DeVito

Regulatory Specialist

NRG Power Marketing Inc.

c/o NRG Energy

10 Southpond Circle

Cheshire, CT 06410

Joel Gordon (Alt)

NRG Power Marketing Inc.

7 Steeple Lane

Amherst, NH 03031

Theodore G. Garill General Manager

Pascoag Fire District Electric Department

55 South Main Street

P.O. Box 107

Pascoag, RI 02859-0107

Roger W. Bacon (Alt)

Pascoag Fire District Electric Department

c/o Mass. Municipal Wholesale Electric Co.

Moody Street P.O. Box 426

Ludlow, MA 01056-0246

Diane Dillman Manager

Paxton Municipal Light Department

578 Pleasant Street Paxton, MA 01612-1365

Roger W. Bacon (Alt)

Paxton Municipal Light Department

c/o Mass. Municipal Wholesale Electric Co.

Moody Street P.O. Box 426

Ludlow, MA 01056-0246

Ronald Tabroff

Supervisor, Electrical Engineer

Peabody Municipal Light Plant

201 Warren Street, Ext. Peabody, MA 01960-4208

Roger W. Bacon (Alt)

Peabody Municipal Light Plant

c/o Mass. Municipal Wholesale Electric Co.

Moody Street P.O. Box 426

Ludlow, MA 01056-0246

Sarah Barpoulis

Senior Vice President

PG&E Energy Trading - Power, LP

7500 Old Georgetown Rd. Bethesda, MD 20816

January 21, 2003

Philip C. Smith (Alt) Director-Operations N.E.

PG&E Energy Trading - Power, LP

c/o PG&E National Energy Group

One Bowdoin Square Boston, MA 02114-2910

Benson Caswell

President

PowerOptions, Inc.

99 Summer Street

Boston, MA 02110-1240

Douglas Stevenson (Alt)

PowerOptions, Inc.

Rte. 1, P.O. Box 398

Wayne, ME 04284

Thomas E. Atkins

Executive Vice President

Power Development Company, LLC

440 Commercial Street Boston, MA 02109

Ken Roberts Jr. (Alt)

Power Development Company, LLC

2 Rockwell Street Niantic, CT 06357

John F. Brodbeck

Energy Consultant

PPL EnergyPlus, LLC

2 North Ninth Street

Allentown, PA 18101-1179

Garrith W. Kirkner (Alt)

PPL EnergyPlus, LLC

2 North Ninth Street

Allentown, PA 18101-1179

Robert A. Weishaar, Jr.

Counsel

Praxair, Inc.

c/o McNees Wallace & Murick LLC 777 North Capitol Street, N.E.

Suite 401

Washington, DC 20002

Jennifer Hunsperger (Alt) Account Manager, Eastern Region

Praxair, Inc. 126 Windham Drive

Langhorne, PA 10947

Debra M. Bateman

Providence Energy Services, Inc.

Mail Stop GENWT21

East Office

56 Exchange Terrace Providence, RI 02903

James M. Stephens (Alt)

Providence Energy Services, Inc.

56 Exchange Terrace Providence, RI 02903

Howard A. Fromer Director – Market Policy

PSEG Energy Resources & Trade LLC

c/o PSEG Power New York Inc. 130 Washington Avenue

Albany, NY 12210

Daniel P. Ciullo (Alt)

Marketing Manager – New England

 $PSEG\ Energy\ Resources\ \&\ Trade\ LLC$

80 Park Plaza – T19 Newark, NJ 07102-4194

Leonard D. Rucker General Manager

Reading Municipal Light Department

230 Ash Street P.O. Box 150

Reading, MA 01867-0250

Vincent F. Cameron Jr. (Alt)

Assistant General Manager - Energy and

Operations

Reading Municipal Light Department

230 Ash Street

Reading, MA 01867-0250

January 21, 2003

Scott Cline

Director, Asset Management and Regulatory

Affairs

Reliant Energy Services, Inc.

1111 Louisiana, 9th floor Houston, TX 77002

Guinette Haas (Alt)

Eastern Region, Asset Management

Reliant Energy Services, Inc.

1111 Louisiana, 9th floor Houston, TX 77002

James T. Rodier

Robert E. McLaughlin Trust

1500A Lafayette Road, No. 112

Portsmouth, NH 03801

G. Robert Merry

Manager

Rowley Municipal Light Plant

47 Summer Street Rowley, MA 01969

Roger W. Bacon (Alt)

Rowley Municipal Light Plant

c/o Mass. Municipal Wholesale Electric Co.

Moody Street P.O. Box 426

Ludlow, MA 01056-0246

Schaefer, Marc

c/o SESCO, Inc.

2 Elmwood Drive

Rutland, VT 05701

Thomas Halleran

Sempra Energy Trading Group

58 Commerce Road Stamford, CT 06902

Mark Magyar (Alt)

Sempra Energy Trading Group

c/o Financial Management Group 325A Southbridge Street

Auburn, MA 01501

Thomas R. Josie General Manager

Shrewsbury Electric Light Plant

100 Maple Avenue

Shrewsbury, MA 01545-5398

Roger W. Bacon (Alt)

Shrewsbury Electric Light Plant

c/o Mass. Municipal Wholesale Electric Co. Moody Street P.O. Box 426

Ludlow, MA 01056-0246

Silkman, Richard

76 Main Street

Yarmouth, ME 04096

Mary J. Healey

State of Connecticut, Office of Consumer

Counsel

Ten Franklin Square New Britain, CT 06051

Nancy A. Harnick (Alt)

State of Connecticut, Office of Consumer

Counsel

Ten Franklin Square New Britain, CT 06051

Julie Hasham

Policy Development Specialist

State of Maine, Office of the Governor

38 State House Station Augusta, ME 04333-0038

Stephen G. Ward (Alt)

Public Advocate

State of Maine, Office of the Governor

112 State House Station Augusta, ME 04333

Kenneth S. Stambler

Assistant Vice President, Energy Supply

SmartEnergy.com, Inc.

200 Unicorn Park

Woburn, MA 01801

January 21, 2003

Tony Lopez-Lopez (Alt)

Chief Energy Operations Officer

SmartEnergy.com, Inc.

200 Unicorn Park

Woburn, MA 01801

Wayne D. Doerpholz

Manager

South Hadley Electric Light Department

85 Main Street

South Hadley, MA 01075-2706

Roger W. Bacon (Alt)

South Hadley Electric Light Department

c/o Mass. Municipal Wholesale Electric Co.

Moody Street

P.O. Box 426

Ludlow, MA 01056-0246

James Daly

Sprague Energy Corp.

Two International Drive

Portsmouth, NH 03801-6809

Jamie Cote (Alt)

Sprague Energy Corp.

Two International Drive

Portsmouth, NH 03801-6809

John Kilgo Jr.

Manager

Sterling Municipal Light Department

50 Main Street

Sterling, MA 01564-2129

Roger W. Bacon (Alt)

Sterling Municipal Light Department

c/o Mass. Municipal Wholesale Electric Co.

Moody Street

P.O. Box 426

Ludlow, MA 01056-0246

Matthew Morais

Strategic Energy Ltd.

Two Gateway Center

9th Floor

Pittsburgh, PA 15222

Michael Swider (Alt)

Strategic Energy Ltd.

33 Sleeper St., Apt. #303

Boston, MA 02210

R. Scott Whittemore

Manager Energy Services and Planning

Taunton Municipal Lighting Plant

55 Weir Street

P.O. Box 870

Taunton, MA 02780-0870

Brian Forshaw

Taunton Municipal Lighting Plant

c/o Connecticut Municipal Electric Energy

Cooperative

30 Stott Avenue

Norwich, CT 06360-1535

Roger L. Buck

Executive Director

TEC-RI

7 Madeline Drive

Newport, RI 02840-1714

Michael Hoffer (Alt)

Vice Chairman Exec. Board

TEC-RI

7 Madeline Drive

Newport, RI 02840-1714

Gerald P. Skelton

Manager

Templeton Municipal Light Plant

School Street

Baldwinville, MA 01436

January 21, 2003

Roger W. Bacon (Alt)

Templeton Municipal Light Plant

c/o Mass. Municipal Wholesale Electric Co.

Moody Street

P.O. Box 426

Ludlow, MA 01056-0246

Jeffrey Bergman

Utilities Manager

Texas Instruments

MS 10-04

34 Forest Street

Attleboro, MA 02703

Mark Sestak (Alt)

Facilities Engineer

Texas Instruments

MS 10-04

34 Forest Street

Attleboro, MA 02703

Roger Borghesani (Alt)

Chairman

The Energy Consortium

24 Hastings Road

Lexington, MA 02421-6807

Mark T. Ladrow

Marketing Manager

Tractebel Energy Marketing, Inc.

15 New England Executive Park

Suite 205

Burlington, MA 01803

Chris Bergamini

Power Trader

TransAlta Energy Marketing (U.S.) Inc.

151 West Street

Suite 300

Annapolis, MD 21401

Dan LoBue

Power Trader

TransAlta Energy Marketing (U.S.) Inc.

151 West Street

Suite 300

Annapolis, MD 21401

William C. Taylor Vice President

TransCanada Power Marketing Ltd.

110 Turnpike Road

Suite 203

Westborough, MA 01581

Michael E. Hachey (Alt)
Director, Power Marketing

TransCanada Power Marketing Ltd.

110 Turnpike Road

Suite 203

Westborough, MA 01581

Michael S. Grim

Director, Markets & Regulatory

TXU Energy Trading Company LP

1717 Main Street, Suite 17000

Dallas, TX 75201

Brian Agree (Alt)

TXU Energy Trading Company LP

1717 Main Street

Suite 2000

Dallas, TX 75201

Kevin M. Foley General Manager

UAE Lowell Power LLC

2 Tanner Street Lowell, MA 01852

Teri M. Pagano (Alt)

UAE Lowell Power LLC

c/o United American Energy Corp.

50 Tice Boulevard

Woodcliff Lake, NJ 07677

James Steffes

UBS AG

c/o UBS Warburg Energy LLC 1500 Louisiana Street

Houston, TX 77002

January 21, 2003

John Llodra (Alt)

UBS AG

c/o UBS Warburg Energy LLC

1500 Louisiana Street

Houston, TX 77002

Deborah Donovan

Union of Concerned Scientists

c/o Synapse Energy Economics, Inc.

22 Pearl Street

Cambridge, MA 02139

Paul R. Peterson

Union of Concerned Scientists

c/o Synapse Energy Economics, Inc.

22 Pearl Street

Cambridge, MA 02139

Robert T. Gagliardi

Director, Strategic Policy

The United Illuminating Company

157 Church Street - 1-15D

P.O. Box 1564

New Haven, CT 06506-0901

Dennis Hrabchak (Alt)

Vice President Regulatory Policy

The United Illuminating Company

157 Church Street-15th Floor

P.O. Box 1564

New Haven, CT 06506-0901

David K. Foote

Vice President

Unitil Corporation Participant companies

6 Liberty Lane West

Hampton, NH 03842-1720

David B. Doskocil (Alt)

Team Leader

Unitil Corporation Participant companies

6 Liberty Lane West

Hampton, NH 03842-1720

Thomas N. Wies

Vice President & General Counsel

Vermont Electric Power Company, Inc.

366 Pinnacle Ridge Road Rutland, VT 05701

Gerald Spring (Alt)

Vermont Electric Power Company, Inc.

366 Pinnacle Ridge Road Rutland, VT 05701

William J. Gallagher General Manager

Vermont Public Power Supply Authority

P.O. Box 298

Waterbury Center, VT 05677

Brian Evans-Mongeon (Alt)

Manager of Power Supply & Marketing Service

Vermont Public Power Supply Authority

P.O. Box 298

Waterbury Center, VT 05677

Ron Armstrong

Virginia Electric and Power Company

5000 Dominion Blvd. 3N Glen Allen, VA 23060

Robert LaRochelle (Alt)

Virginia Electric and Power Company

5000 Dominion Blvd. 3N Glen Allen, VA 23060

William J. Wallace

Manager

Wakefield Municipal Gas and Light Department

9 Albion Street P.O. Box 190

Wakefield, MA 01880-0390

Mark Noyes

WebGen Systems Inc.

41 William Linsky Way Cambridge, MA 02142

January 21, 2003

Paul Taglianetti (Alt) WebGen Systems Inc.

41 William Linsky Way

Cambridge, MA 02142

Roger W. Bacon (Alt)

Wakefield Municipal Gas and Light

Department

c/o Mass. Municipal Wholesale Electric Co.

Moody Street P.O. Box 426

Ludlow, MA 01056-0246

Joyce M. D. Wood

Wellesley Municipal Lighting Plant

c/o Onway Lake Consulting

11 Forest Road

Raymond, NH 03077

John Scirpoli

Manager

West Boylston Municipal Lighting Plant

4 Crescent Street

West Boylston, MA 01583-1310

Roger W. Bacon (Alt)

West Boylston Municipal Lighting Plant

c/o Mass. Municipal Wholesale Electric Co.

Moody Street P.O. Box 426

Ludlow, MA 01056-0246

Daniel Golubek (Alt)

Manager

Westfield Gas and Electric Light Department

100 Elm Street

Westfield, MA 01085-2907

Roger W. Bacon (Alt)

Westfield Gas and Electric Light Department

c/o Mass. Municipal Wholesale Electric Co.

Moody Street P.O. Box 426

Ludlow, MA 01056-0246

January 21, 2003

NEPOOL Participants Committee Members and Alternates

David K. Singer
Williams Energy Marketing & Trading
Company
One Williams Center
Tulsa, OK 74172

Dennis Keener (Alt)
Williams Energy Marketing & Trading
Company
One Williams Center
Mail Drop 35-7
P.O. Box 3448
Tulsa, OK 74101-9567

Tim Charette Power Marketing Executive **WPS Energy Services Inc.** 1242 Lower Lyndon Street Caribou, ME 04736

Ed Howard (Alt) Power Marketing Executive **WPS Energy Services Inc.** 1242 Lower Lyndon Street Caribou, ME 04736

ATTACHMENT 6

New England Governors and Utility Regulatory and Related Agencies

January 21, 2003

Connecticut

The Honorable John G. Rowland

State Capitol

210 Capitol Ave.

Hartford, CT 06106

Connecticut Department of Public Utility

Control

10 Franklin Square

New Britain, CT 06051-2605

<u>Maine</u>

The Honorable John Balducci

One State House Station

Rm. 236

Augusta, ME 04333-0001

Maine Public Utilities Commission

State House, Station 18

242 State Street

Augusta, ME 04333-0018

Massachusetts

The Honorable Mitt Romney

Office of the Governor

Rm. 360 State House Boston, MA 02133 Massachusetts Department of Telecommunications

and Energy

One South Station Boston, MA 02110

New Hampshire

The Honorable Craig Benson

State House Rm. 208

Concord, NH 03301

New Hampshire Public Utilities Commission

8 Old Suncook Road - Building #1

Concord, NH 03301

Rhode Island

The Honorable Donald Carcieri

State House Rm. 143 Providence, RI 02903 Rhode Island Public Utilities Commission

89 Jefferson Boulevard Warwick, RI 02888

<u>Vermont</u>

The Honorable James Douglas

109 State Street

Montpelier, VT 05609

Vermont Public Service Board

112 State Street, Drawer 20 Montpelier, VT 05620-2701

New England Governors and Utility Regulatory and Related Agencies

January 21, 2003

Michael H. Dworkin, President New England Conference of Public Utilities Commissioners, Inc. c/o Chairman Dworkin, Vermont Public Service Board 112 State Street, Drawer 20 Montpelier, VT 05620-2701

Amy Ignatius, Esq.
Executive Director
New England Conference of
Public Utilities Commissioners, Inc.
One Eagle Square
Suite 514
Concord, NH 03301

Harvey L. Reiter, Esq.
Counsel for New England Conference of Public Utilities Commissioners, Inc. c/o MORRISON HECKER
1150 18th Street, N.W., Suite 800
Washington, DC 20036-3816

Power Planning Committee New England Governors' Conference, Inc. 76 Summer Street, 2nd Floor Boston, MA 02110

ATTACHMENT 7

UNITED STATES OF AMERICA FEDERAL ENERGY REGULATORY COMMISSION

Thew Eligiand I owel I out and 150 frew Eligiand Inc I Docket No. ER02-2550-	New England Power Pool and ISO New England Inc.,	Docket No. ER02-2330-
--	--	-----------------------

NOTICE OF FILING

(January , 2003)

Take notice that on January 21, 2003, the New England Power Pool ("NEPOOL") Participants Committee and ISO New England Inc., ("ISO-NE") submitted their Report of Compliance in response to the requirements of the Commission's December 20, 2002 order in New England Power Pool, 101 FERC ¶ 61,344 (2002).

The Participants Committee states that copies of these materials were sent to the New England state governors and regulatory commissions and the Participants in NEPOOL.

Any person desiring to intervene or to protest this filing should file with the Federal Energy Regulatory Commission, 888 First Street, N.E., Washington, D.C. 20426, in accordance with Rules 211 and 214 of the Commission's Rules of Practice and Procedure (18 CFR 385.211 and 385.214). Protests will be considered by the Commission in determining the appropriate action to be taken, but will not serve to make protestants parties to the proceeding. Any person wishing to become a party must file a motion to intervene. All such motions or protests should be filed on or before the comment date, and, to the extent applicable, must be served on the applicant and on any other person designated on the official service list. This filing is available for review at the Commission or may be viewed on the Commission's web site at http://www.ferc.gov, using the "FERRIS" link. Enter the docket number excluding the last three digits in the docket number filed to access the document. For assistance, please contact FERC Online Support at FERCOnlineSupport@ferc.gov or toll-free at (866)208-3676, or for TTY, contact (202)502-8659. Protests and interventions may be filed electronically via the Internet in lieu of paper; see 18 CFR 385.2001(a)(1)(iii) and the instructions on the Commission's web site under the "e-Filing" link. The Commission strongly encourages electronic filings.

Comment Date: February , 2003

Magalie Roman Salas Secretary

CERTIFICATE OF SERVICE

I hereby certify that I have this day served the foregoing document upon each person designated on the official service list compiled by the Secretary in this proceeding.

Dated at Hartford, Connecticut this 21st day of January, 2003.

Steven K. Steinmetz Day, Berry & Howard LLP CityPlace I Hartford, CT 06103-3499

Tel: (860) 275-0266 Fax: (860) 275-0343

E-mail: sksteinmetz@dbh.com